

Citation: Ke W, Reed JN, Yang C, Higgason N, Rayyan L, Wählby C, et al. (2021) Genes in human obesity loci are causal obesity genes in *C. elegans.* PLoS Genet 17(9): e1009736. <u>https://doi.org/</u> 10.1371/journal.pgen.1009736

Editor: Gregory P. Copenhaver, The University of North Carolina at Chapel Hill, UNITED STATES

Received: May 27, 2021

Accepted: July 23, 2021

Published: September 7, 2021

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pgen.1009736

Copyright: © 2021 Ke et al. This is an open access article distributed under the terms of the <u>Creative</u> <u>Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its <u>Supporting</u> Information files. Figures that have associated Raw data include: Figs <u>1E</u>, <u>2B</u>, <u>2C</u>, <u>2F</u>, <u>3A</u>, <u>3B</u>, <u>3C</u>, <u>4B</u>, <u>4C</u>, <u>4E</u>, <u>4F</u>, <u>4G</u>, <u>4H</u>, <u>4I</u>, <u>4J</u>, <u>4K</u>, <u>4L</u>, <u>5B</u>, <u>5C</u>, <u>5D</u>, <u>5E</u>, <u></u> **RESEARCH ARTICLE**

Genes in human obesity loci are causal obesity genes in *C. elegans*

Wenfan Ke¹, Jordan N. Reed², Chenyu Yang¹, Noel Higgason¹, Leila Rayyan¹, Carolina Wählby³, Anne E. Carpenter⁴, Mete Civelek^{2,5}, Eyleen J. O'Rourke^{1,6}*

 Department of Biology, College of Arts and Sciences, University of Virginia, Charlottesville, Virginia, United States of America, 2 Department of Biomedical Engineering, School of Engineering and Applied Science, University of Virginia, Charlottesville, Virginia, United States of America, 3 Department of Information Technology and SciLifeLab, Uppsala University, Uppsala, Sweden, 4 Imaging Platform, Broad Institute of MIT and Harvard, Cambridge, Massachusetts, United States of America, 5 Center for Public Health Genomics, School of Medicine, University of Virginia, Charlottesville, Virginia, United States of America, 6 Department of Cell Biology, School of Medicine, University of Virginia, Charlottesville, Virginia, United States of America

* ejorourke@virginia.edu

Abstract

Obesity and its associated metabolic syndrome are a leading cause of morbidity and mortality. Given the disease's heavy burden on patients and the healthcare system, there has been increased interest in identifying pharmacological targets for the treatment and prevention of obesity. Towards this end, genome-wide association studies (GWAS) have identified hundreds of human genetic variants associated with obesity. The next challenge is to experimentally define which of these variants are causally linked to obesity, and could therefore become targets for the treatment or prevention of obesity. Here we employ highthroughput in vivo RNAi screening to test for causality 293 C. elegans orthologs of human obesity-candidate genes reported in GWAS. We RNAi screened these 293 genes in C. elegans subject to two different feeding regimens: (1) regular diet, and (2) high-fructose diet, which we developed and present here as an invertebrate model of diet-induced obesity (DIO). We report 14 genes that promote obesity and 3 genes that prevent DIO when silenced in C. elegans. Further, we show that knock-down of the 3 DIO genes not only prevents excessive fat accumulation in primary and ectopic fat depots but also improves the health and extends the lifespan of C. elegans overconsuming fructose. Importantly, the direction of the association between expression variants in these loci and obesity in mice and humans matches the phenotypic outcome of the loss-of-function of the C. elegans ortholog genes, supporting the notion that some of these genes would be causally linked to obesity across phylogeny. Therefore, in addition to defining causality for several genes so far merely correlated with obesity, this study demonstrates the value of model systems compatible with in vivo high-throughput genetic screening to causally link GWAS gene candidates to human diseases.

6B, 6C, 6D, 7A, 7B, 7C, 7D, 7E and S1E, S3A, S3B, S4A, S4B, S4C, S4D, S4E Figs. Raw images are available from the corresponding author upon request. All reagents presented in this study and necessary to replicate the findings are available from the corresponding author upon request.

Funding: M.C is funded by National Institutes of Health (DK118287) A.E.C. is funded by National Institutes of Health (GM122547) E.J.O'R. is funded by Pew Charitable Trusts (Biomedical Scholars Award), National Institutes of Health (DK087928), W. M. Keck Foundation, and Jeffress Trust Award The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. NIH (DK118287) supported the salaries of M.C. and J.N.R. NIH (GM122547) supported the salary of J.N.R. NIH (GM122547) supported the salary of A.E.C. All other funders supported salaries of E.J.O'R., W.K., L.R.

Competing interests: The authors have declared that no competing interests exist.

Author summary

Human GWAS have identified hundreds of genetic variants associated with human obesity. The genes being regulated by these variants at the protein or expression level represent potential anti-obesity targets. However, for the vast majority of these genes, it is unclear whether they cause obesity or are coincidentally associated with the disease. Here we use a high-throughput genetic screening strategy to test *in vivo* in *Caenorhabditis elegans* the potential causal role of human-obesity GWAS hits. Further, we combined the results of the genetic screen with analyses of mouse and human GWAS databases. As a result, we present 17 genes that promote or prevent *C. elegans* obesity, and the early onset of organismal deterioration and death associated with obesity. Further, the sign of the correlation between the expression levels of the human genes and their associated clinical traits matches, for the most part, the phenotypic effects of knocking down these genes in *C. elegans*, suggesting conserved causality and pharmacological potential for these genes.

Introduction

Obesity is a major risk factor for serious comorbidities including cardiovascular disease (CVD), type 2 diabetes, hypertension, stroke, neurodegenerative disease, and certain cancers [1,2]. In the past decade, obesity-derived comorbidities caused more than 4 million deaths per year and cost an average of 4 years of life lost worldwide [3–5]. The increased prevalence of obesity and extreme obesity could lead to a global average reduction in life expectancy in the next decade [6,7]. Therefore, there is an urgent need to improve the toolset to reduce the burden of obesity [8–10]. Identifying more effective points of intervention to treat obesity is difficult because obesity is a complex disease influenced by various interacting factors including socioeconomic status, physical activity, eating habits, microbiota, and multiple genes acting in multiple tissues [11].

On one side, the rapid increase in the prevalence of obesity observed in recent decades has been attributed to an "obesogenic environment, which offers ready access to high-calorie foods but limits opportunities for physical activity" (www.CDC.gov). Studies aimed to elucidate the role of carbohydrates in obesity suggested that high sugar consumption could be the main factor behind the alarming increase in the incidence of obesity and metabolic syndrome. More specifically, increased fructose intake from high-fructose corn syrup was found to be highly correlated with the increase in the incidence of obesity observed in the past 40 years [12–14]. Compared to glucose, fructose consumption is more effective at promoting excessive visceral fat accumulation [15] [16]. In line with this, in both humans and rodents, high-fructose diets (HFrD) are functionally linked to the general metabolic dysregulation associated with obesity, also known as metabolic syndrome [17–20].

On the other side, even when we share an obesogenic environment not everyone becomes obese, implying that a sizable portion of the variation in weight among adults would be due to genetic factors. Further, protein-coding genes whose sole inactivation is sufficient to prevent, ameliorate, or revert obesity equate to promising druggable targets. Genome-wide association studies (GWAS) are one of the most promising approaches to identify these gene targets because GWAS search for genetic variants associated with obesity in human populations living in their daily environments (as opposed to animal models reared in controlled conditions). Once identified, the genetic variants can be mapped to genes that may increase or decrease the likelihood of obesity occurrence [21]. Currently, there are over 90 GWAS, which associate 2,537 SNPs with different obesity metrics including body mass index (BMI), waist to hip ratio (WHR), WHR adjusted for BMI, body fat distribution, and/or body fat percentage [22].

Almost 90% of these variants are found in non-coding regions of the genome [23]; hence, very few variants have been mapped to genes. Further, even for those variants that can be reliably mapped to genes, after two decades, only a handful have been causally linked to the disease [24-34]. A major barrier to distinguish statistical association from causal association for the hundreds of loci associated with obesity or diet-induced obesity (DIO) is the lack of *in vivo* systems that enable experimental testing at a reasonable throughput and cost [35].

Genetic screening in *Caenorhabditis elegans* has effectively identified drug targets for human diseases ranging from depression (*e.g.*, Prozac) [36] to metabolic disease (*e.g.*, metformin) [37]. As the first, and only, model organism enabling whole-genome systemic RNA interference (RNAi) *in vivo* through feeding, *C. elegans* is an ideal system for high throughput identification of gene function [38,39]. *C. elegans* is evolutionarily distant from humans. Nevertheless, core lipid, sugar, and protein metabolism pathways are conserved between the two species [40]. Regulators such as TOR kinase and AMPK, and transcription factors such as Sterol response element binding protein (SREBP), Peroxisome proliferator-activated receptor gamma (PPAR γ), and Transcription Factor EB (TFEB), similarly control metabolic genes and cellular responses to nutrients in both organisms. Loss of function of such regulators causes similar metabolic dysregulation, such as obesity or resistance to it, in worms and mammalian systems [41–45]. Moreover, in terms of identifying druggable targets, an obesity candidate gene identified in human GWAS whose ortholog is demonstrated to contribute to obesity in *C. elegans* is more likely to be a robust anti-obesity target across human populations.

As for diet-induced obesity (DIO), a high-glucose DIO model that shows both increased fat accumulation and shortening of lifespan has been previously described for *C. elegans* [46,47]. Less unanimous, though, are previous reports on the effect of fructose supplementation on the *C. elegans* diet. A diet supplemented with 50mg/mL or 100mg/mL of fructose was shown to reduce *C. elegans* healthspan and lifespan by Lodha *et al* [48] and Zheng *et al* [49], respectively. However, fructose at doses of 10 and 20mg/mL extended *C. elegans*'s lifespan. Further, the obesogenic effect of dietary supplementation of fructose has not been previously demonstrated in *C. elegans*.

In this study, we exploit a high-throughput *in vivo* obesity screen system to test for causality genes significantly associated with obesity in human GWAS. First, we identified 340 candidate genes from published GWAS and built a *C. elegans* RNAi library containing 293 worm orthologs of the human-obesity candidate genes. We used our previously developed screening pipe-line [39,50] to perform an RNAi screen for genes whose inactivation alters the fat content of *C. elegans* fed a regular diet (RD). In this screen, we found 14 obesity genes (inactivation leads to obesity) and two lean genes (inactivation leads to leanness). We also established a *C. elegans* DIO model by feeding worms a high-fructose diet (HFrD). We show that worms fed a HFrD not only have higher fat content and body size, but also exhibit shortened health and lifespan. Using this DIO model, we identified three human genes whose *C. elegans* orthologs are DIO suppressors (inactivation prevents HFrD-induced obesity). Furthermore, we show that inactivation of the three DIO suppressors also ameliorates the detrimental effects of a HFrD on *C. elegans* health and lifespan. Altogether, this study provides a path to validate human GWAS obesity candidates *in vivo* in a high-throughput manner for future development of pharmacological interventions to reduce the burden of obesity.

Results

Meta-analysis of obesity GWAS variants identifies 207 human obesity candidate genes with orthologs in *C. elegans*

To identify genes that potentially contribute to human obesity, we exploited published metaanalyses of genome-wide association studies (GWAS) of obesity traits [51,52]. Together, three studies report over 1,200 loci associated with an increased body-mass index (BMI), waist/hip ratio (a measure of fat distribution), or other metabolic traits. However, as is standard for these kinds of studies [53-55], the majority of the variants were not linked to genes. Therefore, we searched through the three previous studies for high-quality candidate genes linked to obesity loci. First, work by our group used eQTL analysis of 770 subcutaneous adipose samples from the Metabolic Syndrome in Men (METSIM) study [56] to link about 680 of these loci to 211 obesity gene candidates (S1 Table). We previously found genetic variants that were associated with both the expression of the candidate gene in subcutaneous adipose tissue and with various metabolic GWAS traits. For the present study, we considered a subset of 211 genes that are associated with obesity-related traits only. We performed Mendelian Randomization analyses to confirm that these genes are likely causal for obesity. Secondly, 120 novel gene candidates were taken based on the genomic location from a study that searched for novel loci associated with BMI in a trans-ancestral meta-analysis of 173,430 samples [57] Finally, Chu et al. [58] analyzed 2.6 million SNPs in up to 9,594 women and 8,738 men of European, African, Hispanic, and Asian ancestry, and based on the genomic location, they predicted 11 novel genes as associated with ectopic fat accumulation in the cohort [58], which we added to our candidate list. Together, we extracted 340 novel genes associated with obesity traits in humans from the three studies (S1 Table, [56–58]). Next, using the comparative genomic analysis tool Ortholist2 [59], we defined that 207 out of the 340 human candidate genes (67%) had orthologs in C. elegans (Fig 1A and S1 Table). However, in some cases, a human gene had more than one C. elegans ortholog; therefore, we identified a total of 386 C. elegans orthologs that corresponded to the 207 human gene candidates (Fig 1A and S1 Table). We then moved to use in vivo functional genomics screening to determine which of the human-obesity candidate genes have causal relationships with fat storage in C. elegans.

16 human-obesity gene candidates modulate fat storage in standard-diet fed *C. elegans*

Although we identified 386 *C. elegans* orthologs of 207 genes associated with obesity traits in humans, only 293 of these 386 ortholog genes were available in the Ahringer (original and supplementary) [60] or the Vidal [61] *C. elegans* genome-wide RNAi libraries (Fig 1A and S1 Table). Therefore, we built an RNAi sub-library consisting of these 293 worm genes for screening, which corresponded to 187 human-obesity gene candidates (workflow in S1A Fig). To increase the efficiency of the RNAi knockdown we used the RNAi-hypersensitive mutant *rrf-3* (*pk1426*) as the genetic background for the screen [62], after confirming it has wild type quantity and distribution of fat as made evident by the fat-specific dye Oil Red O (ORO) (S1B Fig). Although the loss of *rrf-3* function makes *C. elegans* hypersensitive to RNAi (including in neurons, [62]), it is important to keep in mind that some gene targets may not be accessible to RNAi even in this genetic background. Further, gene-gene and gene-environment interactions may be altered by mutation of *rrf-3*. As such, the phenotypic outcomes of our screen are conditioned by the choice of genetic background (*rrf-3*), which may alter the phenotype of genes with background-specific phenotypic effects (*e.g.*, only evident in metabolically sensitized backgrounds [63,64] or in the wild type background).

RNAi treatments, fat staining with ORO, and imaging for the screen were performed in duplicate and retested in three independent biological replicates as described in the Materials and Methods section. Using Cellprofiler feature extraction coupled to Cellprofiler Analyst classification of the microphotographs of fat-stained worms, RNAi treatments were categorized as follows: (1) wild type: >50% of the worms in the well showed quantities and distribution of ORO signal indistinguishable from that of animals treated with empty vector RNAi control



Fig 1. Human obesity GWAS loci are causal obesity genes in *C. elegans.* Throughout this figure: Error bars = S.E.M. N = numbers of independent biological replicates. * $p \le 0.05$, ** $p \le 0.01$, *** $p \le 0.001$ and **** $p \le 0.0001$. (A) Summary of the meta-analysis and *C. elegans* RNAi screen results.

GWAS obesity candidate genes from three studies [56–58] were selected for RNAi screen. (**B-D**) Representative images of empty vector RNAi controls (EV) and RNAi-treated worms stained with oil red O (ORO). All images are in the same magnification, scale bar = 200 μ m. Hit genes were categorized into 3 groups: (**B**) Obese without obvious pleiotropies; (**C**) Obese with sterility. Blue boxes denote unusual "fat depletion around the vulva" phenotype; and (**D**) obese with developmental delay. (**E**) Quantification of ORO signal intensity in RNAi-treated worms for all treatments shown in panels B-D. Each data point represents the measurement of a single worm relative to the EV mean. N = 3. Ratio t-test was used to make comparisons between EV and the treatments.

https://doi.org/10.1371/journal.pgen.1009736.g001

(EV); (2) obese: after RNAi treatment >50% of the worms in these populations showed more intensity or broader distribution of the ORO signal than the parallel EV controls, suggesting that the gene is necessary to prevent obesity in animals fed a standard diet; and (3) lean: >50% of the worms in the RNAi-treated population showed lesser intensity or narrower distribution of the ORO signal than the parallel EV controls, suggesting that the gene is necessary to store normal levels of fat in *C. elegans* fed a standard diet. The gene knockdowns that exhibited a consistent lean or obese phenotype in three independent replicates were selected for further validation (Fig 1A and S1 Table).

We identified 23 genes altering fat storage in the primary screen carried out in 96-well plates. To validate the causal role of these genes in C. elegans reared in standard conditions, we retested the RNAi-driven phenotypes in worms grown in 6cm RNAi plates. In the retest, as in the primary screen, animals were treated with RNAi from the L1 stage. In this validation setup, ORO signal per worm was quantitated using ImageJ. We confirmed 13 out of 14 of the primary obese-gene hits: puf-8 (human gene PUM2), fbf-2 (PUM2), glp-1 (NOTCH4, EYS), let-355 (DHX33), mys-1 (KAT8), rpac-19 (POLR1D), gon-1 (ADAMTS9), hlh-2 (TCF12), pop-1 (TCF7L2), zfh-2 (ZFHX3), eif-6 (EIF6), nst-1 (GLN3), and Y46E12BL.2 (RRP12) (Fig 1B, 1C, 1D and 1E and S1 Table). Of these obese genes, knockdown of all but *puf-8* also caused sterility; hence, hereinafter we refer to these hits as sterile genes (Fig 1C). However, the seemingly exceptional phenotype of puf-8 may be due to our binary sterile versus fertile classification. In fact, *puf-8* deficiency has been linked to germline phenotypes including reduced germline, reduced progeny size, and a tumorous germline [65,66]. If confirmed with a null mutant, the quantitative (as opposed to qualitative) fertility phenotype of *puf-8* opens the exciting possibility of using it to more finely dissect the mechanisms determining soma versus germline nutrient allocation. Furthermore, RNAi against gon-1, hlh-2, pop-2, and zfh-2 promoted an unusual sterility phenotype, in which animals showed extensive fat depletion around the vulva (Fig 1C), suggesting a possible change in fat tissue distribution when germline/eggs are lacking that also warrants future study. Distinctively, eif-6, nst-1, and Y46E12BL.2 showed a developmental-delay phenotype (Fig 1D). A link between obesity and infertility, and obesity and compromised growth, has been extensively documented [40,67-69]. It would then be important for future studies to define whether obesity is the cause or consequence of these phenotypes.

The lean hits in the primary screen were *rpt-5* (human gene *PSMC3*), *hsp-4* (*HSPA1B*), *let-767* (*HSD17B12*), and *Y71H10B.1* (*NT5C2*). Similar to the obese genes, the lean genes could also be subclassified. Animals treated with RNAi against *Y71H10B.1* showed body fat reduction without developmental delay (Fig 2A and 2B). By contrast, RNAi-driven inactivation of *rpt-5*, *hsp-4*, and *let-767* led to severe developmental delay (S1C Fig, and S1 Table). As previously reported [70], we observed that body fat content increased with age (S1D Fig). Therefore, we sought to test whether we could uncouple the developmental delay from the fat phenotype by starting the RNAi treatment at the L4 instead of the L1 stage. In animals fed RNAi from the L4 stage, we found: (1) No fat phenotype for *hsp-4* (S1E and S1F Fig), suggesting increased resilience to ER-stress in the adult *C. elegans*; (2) Reduced fat content after inactivation of *let-767* (Fig 2A and 2C), showing that the function of this gene in fat metabolism can be



Fig 2. Knock down of *Y71H10B.1/NT5C2* and *let-767/HSD17B12* promotes learness in *C. elegans*. Throughout this figure: Scale bars = 200µm. Error bars = S.E.M. N = number of independent biological replicates. Statistical significance was assessed via ratio t-test, $*p \le 0.05$, $**p \le 0.01$, $***p \le 0.001$ and $****p \le 0.0001$. (A) Representative images of the body fat content and distribution in worms treated with RNAi against *Y71H10B.1/NT5C2* (from the L1 stage) and *let-767/HSD17B12* (from the L4 stage because *let-767* RNAi from the L1 stage results in severe developmental arrest; S1C Fig). (B-C) ORO signal intensity in RNAi-treated worms for the treatments shown in panel **A.** Each data point represents the measurement of a single worm relative to the EV mean. N = 3. (D) Representative image of the body fat content and distribution in worms. Each data point represents the measurement of a single worm relative to the EV mean. N = 5. (E) ORO signal intensity in *rpt-5* RNAi-treated worms. Each data point represents the measurement of a single worm relative to the EV mean. N = 3. (D) Representative image of the body fat content and distribution in worms. Each data point represents the measurement of a single worm relative to the EV mean. N = 3.

https://doi.org/10.1371/journal.pgen.1009736.g002

uncoupled from its role in development, and suggesting that this gene may impair development via metabolic dysfunction; and (3) Surprisingly, although knockdown of *rpt-5* from the L1 stage impaired fat accumulation, knockdown of this same gene from the L4 stage caused obesity (compare S1C Fig to Fig 2D and 2E), providing an extreme example of context-dependent gene function. Therefore, altogether, we identified 14 obese genes–*puf-8(PUM2)*, *eif-6* (*EIF6*), *fbf-2(PUM2)*, *glp-1(NOTCH4*, *EYS*), *gon-1(ADAMTS9)*, *hlh-2(TCF12)*, *let-355* (*DHX33*), *mys-1(KAT8*), *nst-1(GLN3*), *pop-1(TCF7L2)*, *rpac-19(POLR1D)*, *rpt-5(PSMC3)*, Y46E12BL.2(*RRP12*), *and zfh-2(ZFHX3)*–and 2 lean genes–*Y71H10B.1(NT5C2)* and *let-767* (*HSD17B12*) (Summary in Fig 1A and S1 Table). The 16 confirmed worm hits correspond to 16 unique human genes (S1 Table). The phenotypes in *C. elegans* suggest these 16 genes may have a conserved causal role in obesity. In the absence of monogenic human diseases or knock-out mice lines, we decided to approximate a test of this hypothesis using three publicly available datasets: (1) Gene expression data from adipose tissue from the Hybrid Mouse Diversity Panel [71] (HMDP)–a cohort of well-phenotyped male and female mice fed a high-fat diet; (2) Subcutaneous adipose tissue gene expression data from the METSIM cohort–a thoroughly phenotyped cohort of Finnish men [56]; and (3) the GWAS catalog (www.ebi.ac.uk/gwas/home). With the HMDP and METSIM datasets, we could define whether reduced or increased expression of the human obesity candidate genes (or their mouse orthologs) were associated with obesity. The GWAS catalog enabled us to investigate to what extent the gene may be specifically affecting metabolic disease, as opposed to promoting more general sickness with an indirect impact on body-mass index (BMI) or waist-hip ratio (WHR).

In the HMDP cohort, we found 12 mouse orthologs of the *C. elegans* hits (*Adamts9*, *Dhx33*, and *Kat8* were not measured in this study). Remarkably, 7 out of the 8 genes whose expression negatively correlates with obesity traits in the mouse, also lead to obesity when they are knocked down via RNAi in *C. elegans* (Fig 3A–bolded names in the gene set to the left), suggesting that the mouse variants would lead to loss or reduced function of these genes and that these seven genes would play similar roles in the regulation of fat storage in *C. elegans* and the mouse. Specifically, lower expression of *Pum2*, *Polr1d*, *Notch4*, *Tcf7l2*, *Tcf12*, *Psmc3*, and *Zfhx3* correlates with increased body fat percentage, fat mass, body weight, and total mass in the mouse, which is comparable to knockdown of the orthologs *puf-8*, *fbf-2*, *rpac-19*, *glp-1*, *pop-1*, *hlh-2*, *rpt-5*, and *zfh-2* causing obesity in *C. elegans*. On the other hand, we found *Nt5c2* expression to positively correlate with obesity traits in the mouse, which is in line with the knockdown of the *C. elegans* ortholog–*Y71H10B.1–* leading to leanness in *C. elegans* (Fig 3A–bolded name in the gene set to the right).

To test the hypothesis that the *C. elegans* hit genes would have conserved causal roles in human obesity we performed similar analyses using the METSIM cohort. We found 16 human genes corresponding to 16 C. elegans gene hits in this dataset. However, on one hand, NOTCH4 and EYS are both orthologs of C. elegans glp-1, and on the other hand, puf-8 and fbf-2 are both orthologs of human PUM2. We calculated the degree of association between the expression in adipose tissue of the 16 human genes and clinical traits including body mass index (BMI) and waist-to-hip ratio (WHR). BMI was the most strongly associated trait (Fig 3B). Further, most human genes showed expression associations with BMI that were in the same direction as in C. elegans. Specifically, lower expression of PUM2, POLR1D, ADAMTS9, DHX33, NOTCH4, EYS, TCF7L2, and KAT8 correlates with increased BMI, which is comparable to the obesity phenotype observed in C. elegans after RNAi knockdown of their orthologs puf-8, fbf-2, rpac-19, gon-1, let-355, glp-1, pop-1, and mys-1. On the other hand, higher NT5C2 expression positively correlates with obesity (Fig 3B), which is in line with the lean phenotype observed in C. elegans after RNAi-driven knockdown of the ortholog gene Y71H10B.1. A parsimonious hypothesis derived from these correlations would be that if RNAi against a C. elegans gene leads to obesity, and a low-expression variant of its ortholog gene is associated with increased BMI in humans, then the ortholog of the C. elegans hit gene is likely to have a causal role in human obesity. Although we appreciate the caveats of the premises leading to this hypothesis, including that changes in expression may be compensatory and not causal mechanisms, we present these analyses as an approximation to the likelihood that the human and C. elegans genes have causal roles in obesity in both organisms. Therefore, as for the most part, the expression associations with BMI in humans go in the same direction as they do in mice



Human gene

Fig 3. Gene expression-disease phenotype associations are conserved for most *C. elegans* screen hits. (A) Bi-weight Mid-correlations (median based correlation metric) of mouse gene expression and metabolic traits in adipose tissue of the HMDP cohort. Genes negatively associated with fat mass are shown on the left, and positive associations on the right. Bolded mouse ortholog names indicate that the expression of the mouse and the worm genes are associated with fat storage (fat mass in mice) in the same direction. (B) Association (Beta of association) between human gene expression in the subcutaneous adipose tissue and clinical traits in the METSIM cohort. Genes negatively associated with BMI are shown on the left. Positive associations are shown on the right. Bolded human ortholog gene names indicate that the expression of the human ortholog gene names indicate that the expression of the human ortholog names indicate the same correlation sign is observed in worms, mice, and humans. (C) Number of published metabolic and non-metabolic trait GWAS associations for each human ortholog present in the GWAS catalog. The assumption to interpret these data is that genes with variants mostly associated to metabolic traits would more likely be directly involved in metabolic or metabolic-related functions, whereas genes with GWAS variants in mostly non-metabolic traits are more likely to have pleiotorpic effects including indirect weight loss or gain. Colors distinguish association with metabolic (black) and non-metabolic (grey) traits.

https://doi.org/10.1371/journal.pgen.1009736.g003

and *C. elegans*, we hypothesize that the orthologs of the *C. elegans* hit genes are likely to have conserved causal roles in human obesity.

Lastly, a gene variant can impact fat accumulation rather directly, because the gene is involved in fat metabolism or its regulation, or rather indirectly because the variant causes general sickness. We investigated this by defining to what extent a gene variant affects metabolic-specific or metabolic-related traits as opposed to unrelated disease traits. We queried the GWAS catalog for each human ortholog of the C. elegans hit genes. It is worth noting that the METSIM study has not yet been included in the GWAS catalog, therefore it was excluded from this analysis. Variants near KAT8, TCF7L2, DHX33, POLR1D, HSD17B12, NT5C2, and MTCH2 were significantly associated with BMI. Variants near GNL3, NOTCH4, ADAMTS9, and EIF6 were significantly associated with waist/hip ratio adjusted for BMI. (S1 and S2 Locus-Plots). We found 15 out of the 16 loci associated with any human disease traits (DHX33 is unassociated). Of these, 11 genes were associated with metabolic traits beyond obesity such as insulin resistance, fat distribution, and cardiovascular disease in human GWAS (Fig 3C, black bars). Further, the variants linked to ADAMTS9, HSD17B12, TCF7L2, KAT8, NT5C2, and GLN3 seem to be mostly associated with metabolic phenotypes as opposed to non-metabolic disease traits (Fig 3C, brown labels indicate genes with the largest proportion of metabolic traits), suggesting these genes may have a rather specific role in metabolic health. In summary, we found supportive evidence in mammalian systems of a role in obesity and metabolic syndrome for most of the orthologs of the genes that we causally linked to obesity in C. elegans.

High-fructose diet leads to diet-induced obesity in C. elegans

Excessive dietary intake of fructose has been suggested to be a major driver of the obesity epidemic [72,73], as fructose is the most common additive in industrialized foods [74]. To test the potential contribution of the human GWAS obesity candidates in the development of DIO we established a fructose-driven *C. elegans* model of DIO. We named high-fructose diet (HFrD) the dietary regimen in which worms are grown from the L1 stage in plates of Nematode growth media (NGM) supplemented with 10mg/mL of fructose (1% fructose). We observed that at the adult stage, worms fed a HFrD show a significant increase in body fat content compared to those fed a regular diet (RD) (Fig 4A and 4B). Further, worms fed the HFrD are also larger than worms fed RD (Fig 4C), which is similar to a previous report showing increased body size in worms fed excessive glucose [75].

Adipose is the main fat storage tissue in humans, and adipocytes with increased lipid droplet (LD) number and size are cellular hallmarks of obesity [76]. In *C. elegans*, there are no specialized adipose cells. The primary triglyceride depots are found in the worm's intestinal cells and are contained in LDs sized between $0.5-1.5 \mu m$ [77,78]. To define whether *C. elegans* fed a



Fig 4. A high-fructose diet leads to diet-induced obesity in C. elegans. Throughout this figure: Error bars = S.E.M. N = numbers of independent biological replicates. Unless specified, unpaired nonparametric t-tests were used to assess the significance. $*p \le 0.05$, $**p \le 0.01$, (**p<0.001 and ****p<0.0001. (A) Representative images of the body fat content and distribution in worms fed regular diet (RD) or highfructose diet (HFrD) from the L1 stage, and stained with ORO when day-1 adults. Scale bar = 200µm. (B) Quantification of ORO intensity in worms for fed RD or HFrD as represented in panel A. Each data point represents the measurement of a single worm. N = 3. (C) Body size quantification of worms fed RD or HFrD. Each data point represents the ImageJ-measured length of a single worm. N = 3. (D) Tissuespecific LD phenotype in worms fed RD (top) or HFrD (bottom). Description of the transgenic reporters is depicted in the panel. Scale bar = 20µm. (E) Mean GFP intensity of intestinal LDs per worm, in worms fed RD or HFrD as represented in D. Each data point represents the measurement of a 0.004mm² intestinal square area of a worm. N = 3. (F) Mean intestinal LD size in worms fed RD or HFrD. Each data point represents the size measurement of an individual LD randomly selected from \geq 10 worms from each independent biological replicate. N = 3. (G) Mean GFP intensity of muscle LDs in worms fed RD or HFrD, as shown in D. Each data point represents the measurement of a 0.003 mm² square area of muscle behind the pharynx of a worm. N = 3. (H) Mean GFP intensity of hypodermal LDs in worms fed RD or HFrD, as shown in D. Each data point represents the measurement of a 0.003mm² square area of hypodermis behind the pharynx of a worm. N = 3. (I) Lifespan assessment of worms fed RD or HFrD. N = 1 is shown in this panel, N = 3 summarized in S2 Table. (J) Locomotory capacity measured as body bends per second in worms fed RD or HFrD. N = 3. (K) Locomotory capacity measured as displacement velocity in worms fed RD or HFrD while on the surface of bacteria-free NGM plates. N = 3. (L) Neurolocomotory and sensorial capacity measured in a food race assay as described in Materials and Methods section. Genotypes: GRU101 = worms with panneuronal expression of non-toxic human A β , and GRU102 = worms with pan-neuronal expression of toxic human A β . N = 3.

https://doi.org/10.1371/journal.pgen.1009736.g004

HFrD would show changes in the abundance and/or size of the LDs, we used a LD reporter consisting of Drosophila PLIN1::GFP driven by the intestine-specific promoter *daf-22* [79,80]. In agreement with the mammalian subcellular hallmarks of obesity, we observed an increase in both the overall intensity (Fig 4D and 4E) and the size (Fig 4F) of the LDs in the intestine of HFrD-fed worms. Another common feature of human obesity is the increase in ectopic fat stores (fat outside the primary fat storage tissues) [81], with excess ectopic fat being associated with worse health outcomes [82]. Although C. elegans does not have a dedicated adipose tissue, fat beyond the intestine is observed in mutants with metabolic dysregulation such as the mTORC2-component mutant rict-1, in which fat is additionally observed in muscle and hypodermis [83,84]. To further characterize the distribution of the fat stores in the HFrD worms, we used the Drosophila PLIN1::GFP construct driven by the *unc-54* promoter to express it in the muscle, or driven by the *Y37A1B.5* promoter to express it in the hypodermis [79,80]. We again observed a significant increase in the LD intensity (Fig 4D, 4G and 4H) but no change in LD size (S3A and S3B Fig), in the muscle and hypodermis of worms fed HFrD. These observations suggest excess fat is stored in primary and ectopic tissues in C. elegans, comparable at a subcellular level to what is observed in visceral and ectopic fat depots in humans with obesity.

Obesity is defined by the World Health Organization as abnormal or excessive fat accumulation that presents a higher risk of debilitating co-morbidities and death [85-87]. In C. elegans, excessive fat accumulation is not always associated with deleterious co-morbidities in health. For instance, mutation of the C. elegans insulin receptor (IR) daf-2 leads to excessive fat accumulation [67]. However, daf-2 mutant animals are resistant to age-related decline and long lived [88–90]. Consistently, even though GH receptor knockout (GHRKO) mice have increased fat mass and decreased lean mass, they show improved insulin sensitivity and are long lived [91]. Therefore, to establish an informative model of DIO in C. elegans, it is critical to test whether the increase in fat levels correlates with detrimental effects on health. To this end, we first evaluated the effect of HFrD on overall survival by comparing the lifespan of HFrD-fed worms to the lifespan of the RD-fed worms. HFrD reduced the median lifespan of C. elegans by 69% (Gehan-Breslow-Wilcoxon test), with a 31.49 average Hazard Ratio (Mantel-Haenszel test) (Fig 4I and S2 Table). Next, we conducted healthspan assays as described previously [90]. In the case of locomotory capacity, we found a significant reduction in body bending rate and average velocity in 3-day old worms fed HFrD when compared to worms of the same age fed RD (Fig 4J and 4K). In C. elegans, locomotion defects including impaired body bending and reduced velocity can be caused by reduced proteostasis [92], and in humans, obesity is strongly associated with neurodegenerative diseases characterized by uncurbed protein aggregation [93–95]. Therefore, we hypothesized that high fructose levels in the C. elegans diet may reduce proteostasis, which could in turn lead to an earlier onset of locomotion defects. As previously reported, C. elegans constitutively expressing a toxic form of the human Aβ amyloid (strain GRU102) show an earlier onset of locomotory impairment than worms expressing a non-toxic form of the human Aβ amyloid (strain GRU101), and this reduced locomotory capacity is due to reduced proteostasis in the neuro-locomotory system [92,96]. To test the hypothesis that reduced proteostasis would contribute to the reduced locomotory capacity observed in worms fed HFrD, we fed a regular diet or HFrD to the GRU101 and GRU102 worms. We conducted a "food race" assay as previously described [97]. Briefly, we placed fifty 3-day old adult worms from each condition (GRU102 ± fructose and GRU101 ± fructose) at one end of a 10cm NGM plate. To start the assay, 25µl of E. coli suspension $(OD_{600nm} = 20)$ were placed on the plates on the opposite side of the worms. After 1h, the number of the worms fully or partially within the borders of the mini bacterial lawn was counted. We consistently found fewer worms expressing toxic A β than worms expressing the non-toxic A β in the lawns (Fig 4L), confirming the published detrimental effect of increased protein aggregation on C. elegans locomotion. We also observed HFrD to be sufficient to impair C. elegans locomotion (Fig 4L). Tellingly, the HFrD-derived impairment was not additive to the expression of toxic A β (Fig 4L), suggesting the possibility that HFrD and A β overexpression may share a common mechanism of toxicity, which we hypothesize might be reduced proteostasis. Altogether, we found that overconsumption of fructose in C. elegans evokes several of the hallmarks of human obesity including elevated body fat content in primary and ectopic fat storage tissue subcellularly characterized by increases in the number and size of the LDs, in conjunction with reduced health and lifespan.

Three human-obesity gene candidates cause DIO in C. elegans

We tested the 293 *C. elegans* orthologs of the human obesity GWAS candidate genes for DIO causality by screening for RNAi treatments that led to reduced or no obesity even when worms were fed excessive fructose. The screening set up for this DIO screen is as described for the regular-diet screen (S1A Fig) except that 10mg/mL of fructose (1% fructose) were supplemented to the Nematode growth media plates. After knocking down the 293 *C. elegans* genes, we observed two phenotypic classes: (1) Wild type: after RNAi treatment, worms in these populations showed the larger quantity and broader distribution of ORO signal that characterizes DIO; this was the expected ORO phenotype because the worms were feeding excessive levels of fructose; and (2) DIO suppressors: \geq 50% of worms in an RNAi treatment showed reduced fat content when compared to EV controls, suggesting that the gene normally contributes to HFrD-driven obesity in *C. elegans*. The RNAi treatments that showed a DIO-suppressor phenotype consistently in 3 or more independent biological replicates were selected for further validation.

From the primary DIO screen, we identified eight high-confidence DIO-suppressor genes. Using the same rationale and approach described above (S1A Fig), we retested the eight genes in 6cm NGM plates. We confirmed the DIO-suppressor phenotype for five out of the eight primary hits. From the five DIO suppressors, *pho-1* (human gene *ACP2*) and *Y71H10B.1* had no detrimental effects on development (Fig 5A, 5B and 5C). However, the other three genes–*rpt-5*, *hsp-4*, and *let-767* –caused severe developmental delay (S3C Fig and S1 Table). Starting the RNAi treatments against these genes at the L4 stage, showed that post-developmental inactivation of *rpt-5* and *hsp-4* did not prevent DIO (S3D Fig). On the other hand, L4-knockdown of *let-767* reduced body fat content and body size in worms fed HFrD (Fig 5A, 5D and 5E),



Fig 5. *Y71H10B.1/NT5C2, let-767/HSD17B12,* and *pho-1/ACP2* inactivations prevent diet-induced obesity in *C. elegans.* Throughout this figure: Scale bars = 200 μ m. Error bars = S.E.M. N = number of independent biological replicates. Statistical significance was assessed via ratio t-test, *p \leq 0.05, **p \leq 0.01, ***p \leq 0.001 and ****p \leq 0.0001. (**A**) Representative images of the body fat content and distribution observed in worms fed HFrD while treated with RNAi against the following genes: Left = *pho-1* and *Y71H10B.1* RNAi treatment from the L1 stage. Right = *let-767* RNAi treatment from the L4 stage. (**B** & **D**) ORO signal intensity in worms treated with RNAi as shown in panel A. Each data point represents the ORO intensity values in each worm normalized to the mean ORO intensity value of the EV

control. N = 3. (C & E) Body size of worms treated with RNAi as shown in panel A. Each data point represents the measurement of a worm. Individual body size values were normalized to the mean value of the EV control. N = 3.

https://doi.org/10.1371/journal.pgen.1009736.g005

suggesting that *let-767* would independently modulate fat metabolism and development. Importantly, *let-767* and *Y71H10B.1* were also lean hits in the regular diet screen described above (Fig 2), suggesting a generic function for these genes in promoting fat accumulation. By contrast, *pho-1* knockdown only prevented obesity in worms fed a HFrD, suggesting a specific function in DIO. Together, we identified and confirmed three genes–*pho-1*, *let-767*, and *Y71H10B.1* –whose knockdown prevented the development of obesity in animals fed excessive sugar.

We already described above that expression of the human ortholog of *Y71H10B.1*, *NT5C2*, positively correlates with waist-to-hip ratio and circulating triglycerides (Fig 3B), which is in line with knockdown of *Y71H10B.1* in *C. elegans* promoting leanness in animals fed a regular or high-fructose diet. In a more difficult to interpret case, the expression of the human ortholog of *let-767*, *HSD17B12*, although negatively associated with BMI, is positively associated with free fatty acids, indicating that reduced expression of *HSD17B12* is associated with obesity but also with improved lipid metabolism. Suggesting a similarly complicated role, expression of the mouse ortholog, *Hsd17b12*, is both negatively (*e.g.*, with % body fat) and positively (*e.g.*, glucose and circulating triglycerides) associated with markers of metabolic health (Fig 3B). Therefore, future studies would be necessary to define whether the role of *let-767* in promoting obesity is conserved in mammals. Finally, using the METSIM cohort, we found that the expression of the human ortholog of *pho-1*, *ACP2*, positively correlates with body weight and total triglycerides (S4 Fig), which is in line with our observations showing that knockdown of *C. elegans's pho-1* prevents diet-induced obesity.

DIO suppressors restore *C. elegans* LD physiology in a tissue-specific manner

As described above, HFrD promotes increased accumulation of larger LDs in both primary and ectopic fat depots (Fig 4D). Using the PLIN1::GFP LD-reporter strains, we assessed whether knockdown of the generic lean genes *let-767* and *Y71H10B.1*, and the specific DIO-suppressor gene *pho-1*, would suppress the LD number and/or size phenotypes, in primary and/or ectopic tissues.

C. elegans treated with RNAi against *Y71H10B.1* showed smaller LDs in the intestine (Fig 6A and 6A'), and less overall LD signal in the intestine, muscle, and hypodermis when animals were fed a regular diet (Fig 6A, 6B, 6C and 6D). Similarly, RNAi against *Y71H10B.1* in HfrD fed worms reduced LD size in the intestine (Fig 6A and 6A'), and overall LD signal in the intestine, muscle, and hypodermis (Fig 6A, 6B, 6C and 6D). The results were, hence, in line with a diet-independent role for *Y71H10B.1* in the control of fat storage in *C. elegans*. On the other hand, although *C. elegans* treated with RNAi against *let-767* showed reduced LD size (Fig 6A and 6A'), the treatment did not significantly change LD intensity in intestine or muscle when *C. elegans* were fed RD (Fig 6A, 6B and 6D). By contrast, RNAi against *let-767* appreciably decreased LD-signal intensity in the hypodermis of animals fed RD (Fig 6C and 6D), suggesting that most of the overall loss of fat observed in *let-767* RNAi-treated animals fed regular diet is due to loss of hypodermal fat (Fig 6C and 6D). Distinctively, and showing the complexity of the fat-storage response to diet, *let-767* RNAi-treated worms fed HFrD showed reduced LD size and overall LD signal in the intestine, as well as decreased overall LD signal in the hypodermis (Fig 6A, 6C and 6E). By contrast, LD signal did not change in the muscle when





significance. * $p \le 0.05$, ** $p \le 0.01$, *** $p \le 0.001$ and **** $p \le 0.0001$. Tissue-specific lipid droplet (LD) reporters strains XD3971, XD1875 and XD2458 express *Drosophila PLIN1::GFP* in the intestine (*daf-22* promoter), muscle (*unc-54* promoter), and hypodermis (*Y37A1B.5* promoter), respectively. LD size and abundance assessments as described in the Materials and Methods section. Each data point represents the GFP measurement of a 0.003mm² square area of the tissue. I: intestine, M: muscle, H: hypodermis. N = 3.Unpaired nonparametric t-tests were used to assess the significance. (**A-A'**) LD size in the *C. elegans* intestine is reduced in *Y71H10B.1*, *let-767*, and *pho-1* RNAi-treated animals fed regular or HFrD diet. (**A & D**) LD abundance in the intestine (I) of worms fed regular diet (RD) is reduced in animals treated with RNAi against *Y71H10B.1*, but not *let-767*. *pho-1* increases the LD pool in the intestine. N = 3. (**B & D**) Similar to the intestine, LD abundance in the muscles (M) of *C. elegans* fed regular diet (RD) is reduced in *Y71H10B.1*, but not in *let-767* RNAi-treated animals. *pho-1* RNAi increases the LD pool in the intestine. N = 3. (**C & D**) LD abundance in the hypodermis (H) of *C. elegans* fed regular diet (RD) is reduced in animals treated with RNAi against *Y71H10B.1*, *let-767*, and *pho-1*. N = 3. (**A & E**) LD abundance in the intestine of worms fed a high-fructose diet (HFrD) is reduced by RNAi treatment against *Y71H10B.1* and *let-767*, whereas RNAi against *pho-1* increases the LD pool in the intestine. N = 3. (**B & E**) LD abundance in the muscles of *C. elegans* fed HFrD is reduced in animals treated with RNAi against *Y71H10B.1*, but not *let-767*, and *pho-1*. N = 3.

https://doi.org/10.1371/journal.pgen.1009736.g006

compared to WT worms fed HFrD (Fig 6B and 6E). Therefore, *let-767* contributes to fat accumulation in the hypodermis independent of the diet and in the intestine only in HFrD, whereas it does not contribute to fat accumulation in the muscle of *C. elegans*.

Screening and validation showed that *pho-1* contributes to DIO but not to basal fat accumulation in C. elegans (Figs 5A, S3C and S3D, and S1 Table). Therefore, it was initially surprising to find that regular diet-fed animals treated with RNAi against pho-1 showed smaller LDs in the intestine (Fig 6A and 6A') and reduced overall LD signal in the hypodermis (Fig 6C and 6D). However, simultaneously, pho-1-treated worms showed a subtle increase in overall LD signal in the intestine and muscle (Fig 6A, 6B and 6D). Hence, we hypothesize that smaller LDs in the intestine and overall less LD content in the hypodermis, in conjunction with increased overall LD signal in intestine and muscle, may lead to a net fat signal in pho-1 knockdown animals fed a regular diet that is indistinguishable from WT when scoring whole-body fat using ORO. Similarly surprising, although pho-1 RNAi suppressed DIO, it increased LD overall signal in the intestine of worms fed HFrD (Fig 6A and 6E). However, pho-1 RNAi decreased the overall LD signal in the hypodermis (Fig 6C and 6E). The data then suggest that pho-1 contributes to DIO mainly through enlargement of LDs in the intestine and increasing the abundance of LDs in the hypodermis, but not through changes in muscle LDs. From a different perspective, the observation that knock-down of all 3 lean genes caused a significant reduction in LD signal in the hypodermis of worms fed HFrD (Fig 6C and 6E) suggests that changes in hypodermal lipid content substantially contribute to DIO in C. elegans. Also intriguing, all lean genes suppressed the enlarged body size of worms fed HFrD (Fig 5C and 5E); therefore, it is reasonable to hypothesize that changes in hypodermal fat stores sizably contribute to obesity and the enlarged body size observed in obese C. elegans.

DIO suppressors restore health biomarkers and lifespan of HFrD-fed *C*. *elegans*

DIO reduces *C. elegans* healthspan and lifespan (Fig 4I, 4J, 4K and 4L). Therefore, we next assessed whether the suppressors of DIO would also suppress the shortening of lifespan caused by a HFrD. For the DIO-suppresor genes that do not cause a developmental delay (*pho-1* and *Y71H10B.1*), the RNAi treatment was started on L1 larvae. For the gene causing developmental delay (*let-767*), RNAi was started at the L4 stage. Survival over time was assessed in three independent biological replicates (summarized in <u>S3 Table</u>). Only lifespan reduction or extension with p<0.05 (Gehan-Breslow-Wilcoxon test) in all three independent replicates was considered significant. We found that RNAi against *pho-1*, *let-767*, and *Y71H10B.1* partially suppressed the short lifespan associated with DIO (Fig 7A, 7B and 7C and S3 Table). On the other hand, knockdown of the same genes did not alter *C. elegans*'s lifespan significantly when



Fig 7. The DIO suppressors ameliorate the negative effects of HFrD on *C. elegans* **healthspan and lifespan**. Throughout this figure: Error bars = S.E.M. N = numbers of independent biological replicates. Unless specified, unpaired nonparametric t-tests were used to assess the significance. $*p \le 0.05$, $**p \le 0.01$, $***p \le 0.001$ and $****p \le 0.0001$. (**A-D**) Comparison of the lifespan of worms fed HFrD while treated with empty RNAi vector control (EV) or RNAi against the DIO suppressors: (A) *Y71H10B.1*, (**B**) *let-767*, and (**C**) *pho-1*. N = 1 is depicted in this figure. N = 3 is shown in <u>S3 Table</u> (**D**) Motor capacity of worms fed RD or HFrD and treated with RNAi against the DIO suppressors measured as bends/second after flooding as described in Materials and Methods section. N = 3. (**E**) Spontaneous displacement velocity on the surface of bacteria-free NGM plates of worms fed RD or HFrD and treated with RNAi against the DIO suppressors. N = 3.

https://doi.org/10.1371/journal.pgen.1009736.g007

animals are fed a regular diet (S3 Table). To examine if knockdown of the genes that reduce body fat also restores *C. elegans* health, we evaluated the behavioral response to flooding (swimming response) and spontaneous locomotion of animals with knockdown of the three DIO-suppressor genes (*let-767*, *Y71H10B.1*, and *pho-1*). We found that knockdown of *Y71H10B.1* ameliorated the swimming response (Fig 7D), and that knockdown of all of these genes ameliorated spontaneous locomotion in animals fed HFrD (Fig 7E). These findings support the notion that reducing body fat content can reduce the health and lifespan burdens associated with obesity in *C. elegans*, and provide evidence of the potential value of these genes as pharmaceutical targets.

Discussion

GWAS is a powerful approach to identify loci associated with obesity because large cohorts are sampled across the arc of socioeconomic statuses, physical activity, eating habits, microbiota composition, and genetic diversity. However, connecting the resulting loci to the specific genes that promote or prevent obesity remains challenging. Obstacles include the large candidate-locus lists, defining the specific gene/s influenced by the genetic variants, and the time-consuming nature of molecular, cellular, and physiological studies [98]. For instance, a single decision such as where (*e.g.*, tissue) and when (*e.g.*, developmental stage) to inactivate or hyperactivate a candidate gene may change the outcome of the phenotypic assessment. As a result, comparatively few GWAS loci have been causally linked to the disease of interest. In the case of obesity, there are ~950 loci associated with BMI [51] and ~350 loci associated with waist-hip ratio adjusted BMI [52], yet very few genes have been validated and fewer druggable targets exist. Therefore, methods that can test the causality of GWAS loci quickly and effectively were needed.

In this study, we introduced two complementary approaches to validate potential obesity GWAS genes in a high-throughput manner: bioinformatic analyses of publicly available datasets and high throughput in vivo RNAi screening in the model organism C. elegans. This nematode has been used as a genetically-tractable animal model for uncovering and characterizing the cellular and molecular functions of genes related to complex human diseases such as obesity. Despite significant differences with mammals including lack of specialized adipose tissue for fat storage [99], absence of key mammalian fat regulators such as leptin [100], and a distinct cholesterol metabolism comparing to mammals [101], the core metabolic pathways (e.g., Beta-oxidation) and signals regulating fat build-up and mobilization, fasting, healthspan and lifespan (e.g., Insulin) are present and carry out comparable functions in both C. elegans and mammals [102]. Furthermore, key regulators of metabolism and lifespan were first discovered in C. elegans and then supported by mammalian studies (e.g., DAF-16/FOXO [103]). Nevertheless, there are caveats associated with using RNAi in a nematode model system to test human obesity variants. The caveats range from false negatives due to the distinct biology and anatomy of C. elegans and mammals-as described above-to the fact that some human variants would be gains of function whereas RNAi causes loss of function. Additionally, gene-environment interactions are very complex and they play critical roles on the effect of gene variants. As such, the chosen genetic background may limit the discovery of genes with phenotypic effects only penetrant in specific mutant or wild-type genetic backgrounds. Nevertheless, as demonstrated in this study, combining gene candidate generation from human GWAS with testing causality via in vivo RNAi in C. elegans can aid in identifying genes contributing to complex human diseases such as obesity.

We here reveal 11 novel obese genes (*fbf-2, gon-1, hlh-2, let-355, mys-1, nst-1, pop-1, puf-8, rpac-19, rpt-5 and Y46E12BL.2*) and two novel lean/DIO suppressor genes (*let-767, and pho-1*). We also retrieved three obese genes (*eif-6, glp-1* and *zfh-2*) and one lean/DIO suppressor gene (*Y71H10B.1*) that were previously linked to fat metabolism or obesity.

Among the known fat genes, we previously showed that inactivation of the *C. elegans* Notch receptor, *glp-1*, causes obesity [67]. By contrast, although the human orthologs of *glp-1*, *NOTCH4* and *EYS*, were associated with obesity in GWAS [56,57], they have not been causally linked to the disease. The combination of experimental validation of *glp-1* as an obesity gene in *C. elegans* with our database analyses showing that expression of *NOTCH4* and *EYS* negatively correlates with BMI and other markers of metabolic disease in mice and humans, puts the spotlight on *NOTCH4* and *EYS* as potential anti-obesity targets. However, not all hits from our screen showed a tight phenotypic correlation between worms and mammals. Unlike the

obesity phenotype observed upon whole-body knockdown of *eif-6* in *C. elegans* fed a regular diet, *Eif6* heterozygous mice had reduced body weight gain compared to their wild-type littermates [104]. Similarly, knockdown of *zfh-2* in *C. elegans* leads to obesity in animals fed a regular diet whereas heterozygous mutation of the mouse ortholog, *Zfhx3/Atbf1*, leads to reduced body weight gain [105]. Nevertheless, it is difficult to evaluate the body-weight phenotypes of *Eif6* and *Zfhx3* heterozygous mutant mice because these mutations lead to pleiotropic effects including perinatal mortality, growth retardation, and severe behavioral deficits [104,105], and neither body fat mass nor even circulating triglycerides were reported in the single *in vivo* mouse mutant studies available to date for each of these genes. Further emphasizing the need for future analyses of these genes, reduced expression of *Zfhx3* is strongly associated with increased body fat, fat mass, and markers of metabolic syndrome including insulin sensitivity and high cholesterol in mice (Fig 3A), suggestive of a role in obesity for this gene. Therefore, the apparent contradictory effects of inactivating *eif-6* and *zfh-2* in *C. elegans* and constitutive inactivation of the murine orthologs need further investigation.

In addition to the three previously characterized genes, we identified 14 C. elegans genes not yet causally linked to obesity. The biological functions of most of these genes and their mammalian orthologs have not been fully elucidated. Nonetheless, some have been characterized to a level that enables us to propose hypotheses about their roles in fat storage. For instance, KAT8 - the human ortholog of C. elegans mys-1 - promotes acetylation of fatty acid synthase (FASN), which leads to reduced FASN activity [106]. FASN is the terminal enzyme in de novo lipogenesis; therefore, it is reasonable to hypothesize that reduced KAT8 activity might increase lipogenesis and hence promote obesity, a phenotype that would be in line with the obesity phenotype we describe here for knockdown of the C. elegans ortholog mys-1 and the negative correlation between KAT-8 expression and obesity in humans. Another example, TCF7L2 - the human ortholog of C. elegans pop-1- has been shown as a key regulator of glucose homeostasis. It has been reported that overexpression of a nuclear isoform of Tcf7l2 (mice ortholog) in high-fat diet fed mice improves glucose tolerance, while depletion of Tcf7l2 in mice causes higher glucose levels and impaired glucose tolerance [107,108]. Although the mechanism is not fully elucidated, impaired glucose homeostasis has been strongly associated with obesity in numerous studies [109]. The protective role of Tcf7l2 in glucose homeostasis may suggest a protective role in obesity too. In support of this notion, expression of Tcf7l2 in mice and TCF7L2 in humans negatively correlates with BMI, which is in line with our observation that knockdown of pop-1 promotes obesity in C. elegans. Therefore, future study of the role of TCF7L2 in human obesity is warranted.

Increased sugar intake, especially fructose from high-fructose corn syrup, was suggested to be a leading cause of the current prevalence of obesity [12-14]. Fructose has been reported to have even more detrimental effects than glucose in healthspan [15]. Here we demonstrated that a HFrD not only disrupts lipid homeostasis but causes premature health deterioration a death in *C. elegans*; a result that is in line with two studies showing that dietary supplementation of 5 and 10% fructose reduces lifespan in *C. elegans* [48,49]. Interestingly, one of these studies also showed that feeding 1 or 2% fructose extended *C. elegans* lifespan [49]. This is intriguing because 1% is the dose of fructose we use in the present study, and we observed a dramatic reduction in the lifespan of *C. elegans* subjected to this dietary regimen. However, there are experimental details that may explain the discrepancies. Zheng *et al* supplemented fructose to worm media plates seeded with *E. coli* OP50, while we used plates seeded with *E. coli* HT115. We made the choice of performing our whole study on animals fed the RNAi-competent strain *E. coli* HT115 to maintain the dietary composition consistent between the primary screen and the follow-up studies. The relevance of *E. coli* to *C. elegans* metabolic phenotypes is well documented. Fat storage, healthspan, and lifespan depend on the bacteria

composition [84,110,111]. Further, recent large-scale *E. coli* mutant screens revealed microbial biochemical pathways influencing *C. elegans* fat accumulation [112] and longevity [113]. Based on these studies, we hypothesize that the discrepancies in the *C. elegans* lifespan phenotypes even when fed the same dose of fructose may be due to the use of different *E. coli* strains in the two studies. This is a very interesting observation since little is known about the molecular underpinnings defining the outcomes of different diet-microbiome-host three-way interactions in the development of obesity. Therefore, our *C. elegans* HFrD model can serve as an informative platform to molecularly dissect such complex interactions in future studies.

Importantly, in this study, we define that knockdown of Y71H10B.1, let-767, and pho-1 prevents the development of DIO in C. elegans. The mammalian ortholog of C. elegans Y71H10B.1, named Nt5c2, encodes for a purine nucleotidase. Nt5c2 knockout mice gain less weight when fed a high-fat diet [114], in line with the DIO suppressor phenotype observed in C. elegans treated with RNAi against Y71H10B.1. Furthermore, using adipose tissue gene expression from the METSIM cohort, we defined that the expression of the human ortholog, NT5C2, positively correlates with waist-to-hip ratio and circulating triglycerides, suggesting that inactivation of Y71H10B.1/NT5C2. would have an antiobesity effect in animal species ranging from C. elegans to humans. Similarly, mice carrying a mutation in Acp2, the murine ortholog of the C. elegans DIO suppressor gene pho-1, are smaller and gain less weight than the wild-type controls [115]. Further, again using the METSIM cohort, we found that the expression of the human ortholog, ACP2, positively correlates with body weight and total triglycerides. Interestingly, the function of pho-1-like genes in fat metabolism may be conserved even across kingdoms, as reduction of ACP4 - the predominant ortholog of pho-1 in Arabidopsis-leads to a decrease in total leaf lipids [116]. The congruency of phenotypic effects between inactivation of Y71H10B.1 and pho-1 in C. elegans and their orthologs in mammals supports the notion that these genes would be causally linked to obesity across animals. In addition the similarities strengthen our confidence in the value of our nematode screening approach to defining which GWAS loci are causally linked to obesity.

Given the causal link between obesity and age-related diseases [117], we sought to define whether knocking down of the three DIO suppressor genes–*pho-1*, *let-767*, and *Y71H10B.1–* ameliorates the negative impact of a high-fructose diet on *C. elegans* health and lifespan. We found all 3 DIO suppressors of obesity also partially suppressed the detrimental effects of HFrD on health and lifespan. To approximate whether the orthologs of *pho-1*, *let-767*, and *Y71H10B.1–* may also impact healthspan in humans, we mined the GWAS catalog. One variant (rs7539320-C) that mapped to the human ortholog of *pho-1*, *ACP2*, is associated with HDL-C [118]. HDL-C has been suggested to have a protective role in cardiovascular health [119], which is a leading cause of premature death in humans.

With respect to *let-767*, its mammalian ortholog *Hsd17b12* is a member of the hydroxysteroid dehydrogenase superfamily specifically involved in the synthesis of arachidonic acid [120]. Knock out of *Hsd17b12* in mice causes embryonic death at the E8.5 stage [121], which is in line with the *C. elegans* developmental delay phenotype we report here. As for humans, we show that the expression of *HSD17B12* is positively associated with LDL-C and negatively associated with HDL-C, implying a detrimental effect of *HSD17B12* on health. Further, other 18 variants mapped to *HSD17B12* are associated with diseases including Type-2 diabetes, cardiovascular disease, and coronary artery diseases in the GWAS Catalog, suggesting a role for *HSD17B12* in the modulation of healthspan that is in line with our observation that depletion of *let-767* improves healthspan in *C. elegans*.

Regarding the human ortholog of *Y71H10B.1*, *NT5C2*, there are 77 variants mapped to *NT5C2* associated with a variety of disease-related traits (besides obesity traits), including hypertension, abnormal blood pressure, coronary artery disease, and cardiovascular diseases

in the GWAS Catalog. Moreover, knockout of *Nt5c2* (in mice) not only protects against highfat diet-induced weight gain and adiposity but also improves insulin sensitivity and reduces hyperglycemia [114], all in line with our observation of improved healthspan and lifespan in *C. elegans* treated with RNAi against *Y71H10B.1*.

In summary, we identified 17 genes that promote or prevent *C. elegans* obesity, and the early onset of organismal deterioration and death associated with obesity. By combining the results of in *vivo* studies in *C. elegans* with analyses of mouse and human GWAS databases, we defined that the sign of the correlation between the expression levels of the mouse and human genes and their associated clinical traits matches, for the most part, the phenotypic effects of knocking down these genes in *C. elegans*, suggesting conserved causality and pharmacological potential for these obesity genes.

Materials and methods

C. elegans strains

C. elegans strains N2 (Bristol, UK), NL2099 (*rrf-3(pk1426)*), GRU101 (*gnaIs1 [myo-2p::yfp]*) and GRU102 (*gnaIs2 [myo-2p::YFP+unc-119p::Abeta1-42]*) were obtained from the Caenor-habditis Genetics Center. The tissue-specific lipid droplet reporter strains XD3971 (*xdIs143 [P_{daf-22} PLIN1::GFP rol-6(su1006)]*), XD2458 (*xdIs56[P_{Y37A1B.5} PLIN1::GFP rol-6(su1006)]*), and XD1875 (*xdIs26[P_{unc-54} PLIN1::GFP rol-6(su1006)]*) were kind gifts from Dr. Monica Driscoll and Dr. Xun Huang.

C. elegans media

NGM (nematode growth medium) agar plates were prepared as previously described [122]. RNAi NGM plates were prepared according to the same protocol, except that 1mM IPTG and 50µg/mL carbenicillin (final concentrations) were added to the molten agar. For 96-well RNAi plates, 200µL of the RNAi NGM was added to each well. For high-fructose diet (HFrD) plates, fructose was supplemented to the plates to reach a final concentration of 10mg/mL.

E. coli culturing

For every biological replicate, fresh *E. coli* streaks or library stamps on LB-carbenicillin 50µg/ mL (for RNAi clones) were used. Bacterial cultures were started from single colonies using a sterilized inoculating tip or hedgehog, and grown overnight for <16h in LB carbenicillin 50µg/ mL in the absence of IPTG (or any other additives). For aeration, flasks were shaken at 250 rpm, and 1.2mL deep 96-well plates at 1,000 rpm. For targeted experiments, bacteria were harvested by centrifugation at room temperature and resuspended to OD600nm = 20 in S-buffer (~10X concentrated). For screening, 1.2mL bacterial cultures were resuspended with 20µl of S-buffer. Concentrated *E. coli* were seeded onto NGM or NGM-RNAi plates immediately and never exposed to the cold.

RNAi screen

The RNAi screen for fat regulators was performed following our published protocol[39] with minor modifications briefly described here. The RNAi clones were inoculated in deep 96-well plates containing 1.2mL per well of terrific broth with carbenicillin $50\mu g/mL$, and grown for 12-16h at 37°C and 1,000 rpm. $50\mu l$ of 24x concentrated *E. coli* were seeded onto each well containing 200 μ l of NGM RNAi media. Seventy-five synchronized NL2099 (*rrf-3(pk1426)*) hatchlings were seeded per well and incubated for 60h at 15°C and then 24h at 25°C. Once the worms reached the young adult stage, they were harvested with S-buffer and stained with Oil

Red O (ORO) 0.5%, and imaged as previously described [39]. Individual worms in the microphotographs were identified and >500 parameters extracted using CellProfiler pipelines (cellprofiler.org/wormtoolbox) optimized as previously published [50,123]. The RNAi treatments were classified according to their fat phenotypes using the supervised-machine tool CellProfiler Analyst as previously described [50]. Phenotypic classes in the regular diet and the high-fructose diet screens were defined as described in the Results section.

The regular diet and the HFrD screens were independently repeated three times, and in each of these three experiments, each candidate obesity gene was inactivated twice (experimental replicates). RNAi treatments that consistently showed the same phenotype across experimental and biological replicates were selected for validation. For validation, single colonies of RNAi bacteria were inoculated into 500mL LB and incubated at 37°C, 200 rpm, overnight. Bacteria were then harvested by centrifugation at room temperature and resuspended to $OD_{600nm} = 20$ in S-buffer (~10X concentrated) and seeded onto 6cm RNAi plates previously seeded with water (mock) or fructose 10mg/mL final concentration. Three hundred synchronized NL2099 hatchlings were seeded onto the plates and incubated at 15°C for 60h, followed by 24h incubation at 25°C. Worms were then harvested using S-buffer into 1.5mL tubes and stained with freshly made Oil Red O (ORO) and imaged as previously described [39,50]. ORO intensity was quantified using ImageJ (fiji.sc).

Aging assay

C. elegans were reared on 6cm NGM or NGM RNAi plates seeded with 200µl of 20x concentrated suspensions of fresh overnight cultures of *E. coli* HT115 carrying empty vector RNAi control (L4440 plasmid) or RNAi plasmids targeting the genes of interest. The F1 of well-fed and uncontaminated gravid adults was synchronized using the hypochlorite/NaOH method. Once seeded onto plates, hatchlings were incubated at 15°C for 60h. Once they reached the L4-stage, 30 worms were transferred to two regular diet NGM RNAi plates and two high-fructose diet NGM RNAi plates. The plates were incubated at 25°C for the rest of the experiment. Starting twenty-four hours later (when worms were gravid adults) and until all animals in each population were dead, viability was scored daily by prodding. Worms were classified as dead and removed from the plates if they did not respond for 10 sec to gentle touch with a sterilized platinum wire. Worms were classified as "censored" if they were missing, displayed vulval rupture, or internal hatching. Data were analyzed using the Mantel-Haenszel method to calculate hazard ratio and 95% Cl, and the Gehan-Breslow-Wilcoxon method to calculate the p-value. The experiments were repeated a minimum of 3 independent times for each treatment tested.

Healthspan assays

Healthspan assays were modified from previous publications [90,124]. 3d old adults were harvested from plates into 1.5mL tubes and washed twice with S-buffer. For the body bending assay, 30 3d old worms were added to one well of a 12-well plate (CytoOne) containing 2mL of S-buffer. The plates were kept at RT for 15min prior to imaging. A time-lapse (30s) was recorded using a Zeiss Axio Zoom.v16 dissecting microscope, PlanNeoFluar Z 2.3X/0.57 FWD objective, zoom 7X. The movie was analyzed to count body bends using the ImageJ plug-in wrMTrck(www.phage.dk/plugins/wrmtrck.html). For velocity measurements, 3d-old worms were harvested, washed, resuspended in S buffer, and the concentration of the suspension was estimated. Approximately thirty 3d-old worms were seeded onto an empty 6cm NGM plate (no bacteria). Once the drop of Sbuffer dried, we waited for 5 min at room temperature before starting imaging. Time-lapses (2min at 5fps) were captured using a Zeiss Axio Zoom.v16 dissecting microscope equipped with a PlanNeoFluar Z 2.3X/0.57 FWD objective, at 7X

magnification. Worm displacement velocity was scored using the ImageJ Manual Tracking plugin. For the racing assay, similar to the method described previously [97], 3d-old worms were harvested, washed, resuspended in S buffer, and the concentration of the suspension was estimated. Approximately fifty worms were then transferred to a 10cm NGM plate previously seeded with 25µL of 20X concentrated *E. coli* HT115 placed at the opposite side of the plate from where the worms would be seeded. After 1h, images of the mini lawns were taken using a Zeiss Axio Zoom.v16 dissecting microscope, PlanNeoFluar Z 2.3X/0.57 FWD objective, at 7X magnification. Using the images, the number of worms on the bacterial lawn or touching the lawn were manually counted.

Lipid droplet live imaging and analysis

Day-1 adult worms carrying the lipid-droplet fluorescent reporters (XD1875, XD3971, or XD2458) were harvested, washed twice with S-buffer, paralyzed with 40mg/mL levamisole, and mounted onto agar pads as previously described [125]. Images were captured with a Nikon Eclipse Ti spinning disc confocal microscope, 60X/0.85NA objective, 500ms exposure time, and 80% laser intensity. The images were then analyzed using ImageJ to measure the intensity and the size of the lipid droplets. To measure the intensity, a ROI was defined and the measure function was used to measure the mean intensity per μ m². To measure the LD diameter, random LD was picked and the diameter was measured by using the line and measure function.

Statistics

All statistical analyses were performed using Graphpad Prism. Outliers were detected and removed from analyses using the ROUT method. For absolute intensity measurements, lipid droplet diameter, body size, and velocity, the unpaired nonparametric t-test was used to make single comparisons between each treatment and the mock control. The ratio t-test was used to compare all ratios. Aging statistics performed as described above. Unless otherwise stated, significance is represented as follows: *p \leq 0.05, **p \leq 0.01, ***p \leq 0.001, and ****p \leq 0.0001. Unless otherwise stated, data in this study are presented as mean values +/- SEM. All experiments were performed and quantitated at least three independent times.

Human genetics database

Human gene expression data, clinical trait data, and SNP association data were previously collected and published by the Metabolic Syndrome in Men (METSIM) cohort [56]. This cohort has 770 healthy male subjects with mean BMI = 26.59 ± 3.47 . Gene expression from subcutaneous adipose tissue was measured using Affymetrix U219 microarrays, and extensive clinical phenotyping was performed, including anthropometric traits, insulin sensitivity metrics, and blood metabolite levels. BMI and WHR were both reported. Correlations between gene expression and clinical traits were calculated using the biweight midcorrelation, and only significant associations are shown. Associations between SNPs and gene expression were calculated using a linear mixed model as described in Civelek et al [56]. The SNP with the most significant gene expression within 500 kB of the transcriptional start site was designated as the eSNP. We obtained the associations between BMI and WHR from Pulit et al [52]. LocusZoom plots were generated using the BMI and WHRadjBMI summary statistics from Pulit et al. and genotype and gene expression data from the METSIM cohort [56]. Plots were generated using the LocusZoom browser tool.

Supporting information

S1 Fig. Throughout this figure: Scale bars = 200µm, Error bars = S.E.M. N = number of independent biological replicates. Statistical significance was assessed via ratio t-test, $*p \le 0.05$, $**p \le 0.01$, $***p \le 0.001$ and $****p \le 0.0001$. (A) Overview of the workflow of the screen for genes altering fat storage in *C. elegans*. (B) Representative images of the body fat content and distribution in WT(N2) and *rrf-3* RNAi sensitive mutant worms. N = >5. (C) Representative images of the body fat content and distribution as made evident with ORO in worms treated with RNAi against the obesity candidate genes from the L1 stage. Asterisks denote those RNAi treatments that led to developmental delay. (D) Representative images of the age-dependent increases in fat stores as revealed by staining L3, L4, and 1-day adult worms with ORO. N = >5. (E) Retesting of ORO staining in worms treated with *hsp-4* RNAi from the L3 stage showed no effect of *hsp-4* RNAi as represented in panel E. Each data point represents the ORO intensity in one worm. Individual intensity values were normalized to the mean value of the EV RNAi control. N = 3. (PDF)

S2 Fig. Analysis of human orthologs association to BMI. A) Association to BMI in TWINsUK **B**) Correlation with BMI in GTEx. Blue indicates the direction of association aligns with the lean gene in *C. elegans*. Red indicates the direction of association aligns with the obese gene in *C. elegans*.

(PDF)

S3 Fig. Throughout this figure: Error bars = S.E.M. N = numbers of independent biological replicates. Unless specified, unpaired nonparametric t-tests were used to assess the significance. *p≤0.05, **p≤0.01, ***p≤0.001 and ****p≤0.0001. (A) Quantification of muscle LD size in worms fed RD or HFrD. Each data point represents the measurement of random LDs in the images. ≥ 10 worms were measured in each independent biological replicate. N = 3. (B) Quantification of hypodermis LD size in worms fed RD or HFrD. Each data point represents the measurement of a random LD in the images. ≥ 10 worms were measured in each independent biological replicate. N = 3. (C) Body fat content of worms treated with RNAi against all of the DIO suppressors observed in the primary screen. RNAi was initiated from the L1 stage. pho-1 and Y71H10B.1 (Blue font) are the 2 DIO suppressors that cause leanness without developmental delay. (D) Representative images of the body fat content and distribution in worms fed HFrD from the L1 stage but treated with RNAi against rpt-5 and hsp-4 from the L4 stage. No change in fat content was observed in this condition. N = 3. (E) Representative images of the body fat content and distribution in worms treated with RNAi against *pho-1* from the L4 stage. No change in fat content was observed. N = 3(PDF)

S4 Fig. LocusZoom plots of the associations of the SNPs near *ACP2* with (**A**) BMI and (**B**) *ACP2* gene expression in subcutaneous adipose tissue in the METSIM cohort. The effect of the alleles of rs10501321 on (**C**) *ACP2* expression and (**D**) total triglycerides. (**E**) Correlation between BMI and *ACP2* (human *pho-1* ortholog) expression in the METSIM cohort. (PDF)

S1 Table. Tabulated summary of the results of the GWAS meta-analysis and RNAi screen. List of the 340 human genes associated with obesity traits extracted from 3 previous publications: 1) Mete Civelek, et. al, Am J Hum Genet., 2017 [56]; 2) Masato Akiyama, et.al, Nature Genetics, 2017 [57]; and 3) Audrey Chu, et.al, Nature Genetics, 2017 [58]. 207 of these human genes had C. elegans orthologs. Since there is often more than one *C. elegans* ortholog per human gene, 386 *C. elegans* orthologs were actually identified to correspond to the 207 human genes. However, only 293 *C. elegans* genes were screened because there weren't available RNAi clones for 96 of the C, elegans orthologs. 14 obese, 2 generic lean, and 3 DIO-suppressor genes were identified and validated.

(XLSX)

S2 Table. Lifespan analysis of worms fed HFrD versus RD. Summary of 3 repeats of the lifespan analysis of worms fed HFrD versus RD. Hazard ratios were calculated using Log-rank and Mantel-Haenszel methods.

(XLSX)

S3 Table. DIO suppressors extend lifespan of C. elegans fed HFrD. Summary of 3 repeats of the lifespan analysis of worms treated with RNAi constructs versus EV in RD and HFrD conditions. Hazard ratios were calculated using Log-rank and Mantel-Haenszel methods. (XLSX)

S1 LocusPlots. LocusZoom plots of the associations of the variants within 1Mb of the human genes with BMI. (DOCX)

S2 LocusPlots. LocusZoom plots of the associations of the variants within 1Mb of the human genes with WHRadjBMI in genome wide association studies. (DOCX)

S1 Source Data. This source data file includes all raw data and analysis presented in Figs 1E, 2B, 2C, 2F, 3A, 3B, 3C, 4B, 4C, 4E, 4F, 4G, 4H, 4J, 4K, 4L, 5B, 5C, 5D, 5E, 6B, 6C, 6D, 7D, 7E, S1E, S3A, S3B, S4A, S4B, S4C, S4D and S4E. (XLSX)

S2 Source Data. This source data file includes all raw data and analysis of aging experiments, presented in Figs 4I, 7A, 7B, 7C, and S2 and S3 tables. (XLSX)

Acknowledgments

We thank Dr. Monica Driscoll and Dr. Xun Huang for generously sharing the LD reporter strains. We especially thank Nella Solodukhina for helping manage the lab, preparing reagents, and conducting some of the experiments. We acknowledge the 2020 and 2021 BIOL4005 class of the University of Virginia for their help in cross-validating automated scoring.

Author Contributions

Conceptualization: Wenfan Ke, Eyleen J. O'Rourke.

- **Data curation:** Wenfan Ke, Jordan N. Reed, Chenyu Yang, Noel Higgason, Leila Rayyan, Mete Civelek, Eyleen J. O'Rourke.
- **Formal analysis:** Wenfan Ke, Jordan N. Reed, Chenyu Yang, Noel Higgason, Leila Rayyan, Carolina Wählby, Anne E. Carpenter, Mete Civelek, Eyleen J. O'Rourke.

Funding acquisition: Carolina Wählby, Anne E. Carpenter, Mete Civelek, Eyleen J. O'Rourke.

Investigation: Wenfan Ke, Jordan N. Reed, Chenyu Yang, Noel Higgason, Leila Rayyan, Carolina Wählby, Anne E. Carpenter, Mete Civelek, Eyleen J. O'Rourke. **Methodology:** Wenfan Ke, Jordan N. Reed, Carolina Wählby, Anne E. Carpenter, Eyleen J. O'Rourke.

Project administration: Eyleen J. O'Rourke.

Resources: Wenfan Ke, Eyleen J. O'Rourke.

Software: Wenfan Ke, Jordan N. Reed, Carolina Wählby, Anne E. Carpenter, Mete Civelek, Eyleen J. O'Rourke.

Supervision: Mete Civelek, Eyleen J. O'Rourke.

Validation: Wenfan Ke, Chenyu Yang, Noel Higgason, Leila Rayyan, Eyleen J. O'Rourke.

Visualization: Wenfan Ke, Jordan N. Reed, Mete Civelek, Eyleen J. O'Rourke.

Writing - original draft: Wenfan Ke, Jordan N. Reed, Eyleen J. O'Rourke.

Writing – review & editing: Wenfan Ke, Jordan N. Reed, Anne E. Carpenter, Mete Civelek, Eyleen J. O'Rourke.

References

- Poirier P, Giles TD, Bray GA, Hong Y, Stern JS, Pi-Sunyer FX, et al. Obesity and cardiovascular disease: pathophysiology, evaluation, and effect of weight loss: an update of the 1997 American Heart Association Scientific Statement on Obesity and Heart Disease from the Obesity Committee of the Council on Nutrition, Physical Activity, and Metabolism. Circulation. 2006; 113: 898–918. https://doi.org/10.1161/CIRCULATIONAHA.106.171016 PMID: 16380542
- Ndumele CE, Matsushita K, Lazo M, Bello N, Blumenthal RS, Gerstenblith G, et al. Obesity and subtypes of incident cardiovascular disease. J Am Heart Assoc. 2016; 5. <u>https://doi.org/10.1161/JAHA. 116.003921</u> PMID: 27468925
- Flegal KM, Kruszon-Moran D, Carroll MD, Fryar CD, Ogden CL. Trends in obesity among adults in the united states, 2005 to 2014. JAMA. 2016; 315: 2284–2291. https://doi.org/10.1001/jama.2016.6458 PMID: 27272580
- Ogden CL, Carroll MD, Lawman HG, Fryar CD, Kruszon-Moran D, Kit BK, et al. Trends in Obesity Prevalence Among Children and Adolescents in the United States, 1988–1994 Through 2013–2014. JAMA. 2016; 315: 2292–2299. https://doi.org/10.1001/jama.2016.6361 PMID: 27272581
- Ng M, Fleming T, Robinson M, Thomson B, Graetz N, Margono C, et al. Global, regional, and national prevalence of overweight and obesity in children and adults during 1980–2013: a systematic analysis for the Global Burden of Disease Study 2013. Lancet. 2014; 384: 766–781. https://doi.org/10.1016/ S0140-6736(14)60460-8 PMID: 24880830
- 6. Lim SS, Vos T, Flaxman AD, Danaei G, Shibuya K, Adair-Rohani H, et al. A comparative risk assessment of burden of disease and injury attributable to 67 risk factors and risk factor clusters in 21 regions, 1990–2010: a systematic analysis for the Global Burden of Disease Study 2010. Lancet. 2012; 380: 2224–2260. https://doi.org/10.1016/S0140-6736(12)61766-8 PMID: 23245609
- Olshansky SJ, Passaro DJ, Hershow RC, Layden J, Carnes BA, Brody J, et al. A potential decline in life expectancy in the United States in the 21st century. N Engl J Med. 2005; 352: 1138–1145. <u>https:// doi.org/10.1056/NEJMsr043743</u> PMID: 15784668
- Kang JG, Park C-Y. Anti-Obesity Drugs: A Review about Their Effects and Safety. Diabetes Metab J. 2012; 36: 13–25. https://doi.org/10.4093/dmj.2012.36.1.13 PMID: 22363917
- Rodgers RJ, Tschöp MH, Wilding JPH. Anti-obesity drugs: past, present and future. Dis Model Mech. 2012; 5: 621–626. https://doi.org/10.1242/dmm.009621 PMID: 22915024
- Field BCT, Chaudhri OB, Bloom SR. Obesity treatment: novel peripheral targets. Br J Clin Pharmacol. 2009; 68: 830–843. https://doi.org/10.1111/j.1365-2125.2009.03522.x PMID: 20002077
- Yang W, Kelly T, He J. Genetic epidemiology of obesity. Epidemiol Rev. 2007; 29: 49–61. <u>https://doi.org/10.1093/epirev/mxm004 PMID: 17566051</u>
- Gross LS, Li L, Ford ES, Liu S. Increased consumption of refined carbohydrates and the epidemic of type 2 diabetes in the United States: an ecologic assessment. Am J Clin Nutr. 2004; 79: 774–779. https://doi.org/10.1093/ajcn/79.5.774 PMID: 15113714

- Marriott BP, Olsho L, Hadden L, Connor P. Intake of added sugars and selected nutrients in the United States, National Health and Nutrition Examination Survey (NHANES) 2003–2006. Crit Rev Food Sci Nutr. 2010; 50: 228–258. https://doi.org/10.1080/10408391003626223 PMID: 20301013
- Johnson RJ, Segal MS, Sautin Y, Nakagawa T, Feig DI, Kang D-H, et al. Potential role of sugar (fructose) in the epidemic of hypertension, obesity and the metabolic syndrome, diabetes, kidney disease, and cardiovascular disease. Am J Clin Nutr. 2007; 86: 899–906. <u>https://doi.org/10.1093/ajcn/86.4.899</u> PMID: 17921363
- Stanhope KL, Schwarz JM, Keim NL, Griffen SC, Bremer AA, Graham JL, et al. Consuming fructosesweetened, not glucose-sweetened, beverages increases visceral adiposity and lipids and decreases insulin sensitivity in overweight/obese humans. J Clin Invest. 2009; 119: 1322–1334. <u>https://doi.org/ 10.1172/JCI37385 PMID: 19381015</u>
- Schwarz J-M, Noworolski SM, Erkin-Cakmak A, Korn NJ, Wen MJ, Tai VW, et al. Effects of dietary fructose restriction on liver fat, de novo lipogenesis, and insulin kinetics in children with obesity. Gastroenterology. 2017; 153: 743–752. https://doi.org/10.1053/j.gastro.2017.05.043 PMID: 28579536
- 17. Lustig RH. Fructose: metabolic, hedonic, and societal parallels with ethanol. J Am Diet Assoc. 2010; 110: 1307–1321. https://doi.org/10.1016/j.jada.2010.06.008 PMID: 20800122
- Nakagawa T, Tuttle KR, Short RA, Johnson RJ. Hypothesis: fructose-induced hyperuricemia as a causal mechanism for the epidemic of the metabolic syndrome. Nat Clin Pract Nephrol. 2005; 1: 80– 86. https://doi.org/10.1038/ncpneph0019 PMID: 16932373
- Dhingra R, Sullivan L, Jacques PF, Wang TJ, Fox CS, Meigs JB, et al. Soft drink consumption and risk of developing cardiometabolic risk factors and the metabolic syndrome in middle-aged adults in the community. Circulation. 2007; 116: 480–488. https://doi.org/10.1161/CIRCULATIONAHA.107.689935 PMID: 17646581
- Toop CR, Gentili S. Fructose Beverage Consumption Induces a Metabolic Syndrome Phenotype in the Rat: A Systematic Review and Meta-Analysis. Nutrients. 2016; 8. <u>https://doi.org/10.3390/</u> nu8090577 PMID: 27657120
- Hindorff LA, Sethupathy P, Junkins HA, Ramos EM, Mehta JP, Collins FS, et al. Potential etiologic and functional implications of genome-wide association loci for human diseases and traits. Proc Natl Acad Sci USA. 2009; 106: 9362–9367. https://doi.org/10.1073/pnas.0903103106 PMID: 19474294
- Buniello A, MacArthur JAL, Cerezo M, Harris LW, Hayhurst J, Malangone C, et al. The NHGRI-EBI GWAS Catalog of published genome-wide association studies, targeted arrays and summary statistics 2019. Nucleic Acids Res. 2019; 47: D1005–D1012. <u>https://doi.org/10.1093/nar/gky1120</u> PMID: 30445434
- Maurano MT, Humbert R, Rynes E, Thurman RE, Haugen E, Wang H, et al. Systematic localization of common disease-associated variation in regulatory DNA. Science. 2012; 337: 1190–1195. https://doi. org/10.1126/science.1222794 PMID: 22955828
- 24. Loh NY, Neville MJ, Marinou K, Hardcastle SA, Fielding BA, Duncan EL, et al. LRP5 regulates human body fat distribution by modulating adipose progenitor biology in a dose- and depot-specific fashion. Cell Metab. 2015; 21: 262–273. https://doi.org/10.1016/j.cmet.2015.01.009 PMID: 25651180
- Small KS, Todorčević M, Civelek M, El-Sayed Moustafa JS, Wang X, Simon MM, et al. Regulatory variants at KLF14 influence type 2 diabetes risk via a female-specific effect on adipocyte size and body composition. Nat Genet. 2018; 50: 572–580. https://doi.org/10.1038/s41588-018-0088-x PMID: 29632379
- Gesta S, Bezy O, Mori MA, Macotela Y, Lee KY, Kahn CR. Mesodermal developmental gene Tbx15 impairs adipocyte differentiation and mitochondrial respiration. Proc Natl Acad Sci USA. 2011; 108: 2771–2776. https://doi.org/10.1073/pnas.1019704108 PMID: 21282637
- Loh NY, Minchin JEN, Pinnick KE, Verma M, Todorčević M, Denton N, et al. RSPO3 impacts body fat distribution and regulates adipose cell biology in vitro. Nat Commun. 2020; 11: 2797. <u>https://doi.org/ 10.1038/s41467-020-16592-z PMID: 32493999</u>
- Lee KY, Yamamoto Y, Boucher J, Winnay JN, Gesta S, Cobb J, et al. Shox2 is a molecular determinant of depot-specific adipocyte function. Proc Natl Acad Sci USA. 2013; 110: 11409–11414. https://doi.org/10.1073/pnas.1310331110 PMID: 23798383
- Ollmann MM, Wilson BD, Yang YK, Kerns JA, Chen Y, Gantz I, et al. Antagonism of central melanocortin receptors in vitro and in vivo by agouti-related protein. Science. 1997; 278: 135–138. <u>https://doi.org/10.1126/science.278.5335.135</u> PMID: 9311920
- Hainer V, Aldhoon Hainerová I, Kunešová M, Taxová Braunerová R, Zamrazilová H, Bendlová B. Melanocortin pathways: suppressed and stimulated melanocortin-4 receptor (MC4R). Physiol Res. 2020; 69: S245–S254. https://doi.org/10.33549/physiolres.934512 PMID: 33094623

- Wang P, Loh KH, Wu M, Morgan DA, Schneeberger M, Yu X, et al. A leptin-BDNF pathway regulating sympathetic innervation of adipose tissue. Nature. 2020; 583: 839–844. https://doi.org/10.1038/ s41586-020-2527-y PMID: 32699414
- 32. Ramos-Molina B, Martin MG, Lindberg I. PCSK1 variants and human obesity. Prog Mol Biol Transl Sci. 2016; 140: 47–74. https://doi.org/10.1016/bs.pmbts.2015.12.001 PMID: 27288825
- Tartaglia LA, Dembski M, Weng X, Deng N, Culpepper J, Devos R, et al. Identification and expression cloning of a leptin receptor, OB-R. Cell. 1995; 83: 1263–1271. <u>https://doi.org/10.1016/0092-8674(95)</u> 90151-5 PMID: 8548812
- Spanswick D, Smith MA, Groppi VE, Logan SD, Ashford ML. Leptin inhibits hypothalamic neurons by activation of ATP-sensitive potassium channels. Nature. 1997; 390: 521–525. <u>https://doi.org/10.1038/</u> 37379 PMID: 9394003
- Korte A, Farlow A. The advantages and limitations of trait analysis with GWAS: a review. Plant Methods. 2013; 9: 29. https://doi.org/10.1186/1746-4811-9-29 PMID: 23876160
- Choy RK, Thomas JH. Fluoxetine-resistant mutants in C. elegans define a novel family of transmembrane proteins. Mol Cell. 1999; 4: 143–152. https://doi.org/10.1016/s1097-2765(00)80362-7 PMID: 10488330
- Wu L, Zhou B, Oshiro-Rapley N, Li M, Paulo JA, Webster CM, et al. An ancient, unified mechanism for metformin growth inhibition in c. elegans and cancer. Cell. 2016; 167: 1705–1718.e13. <u>https://doi.org/ 10.1016/j.cell.2016.11.055</u> PMID: 27984722
- O'Rourke EJ, Conery AL, Moy TI. Whole-animal high-throughput screens: the C. elegans model. Methods Mol Biol. 2009; 486: 57–75. https://doi.org/10.1007/978-1-60327-545-3_5 PMID: 19347616
- Ke W, Drangowska-Way A, Katz D, Siller K, O'Rourke EJ. The Ancient Genetic Networks of Obesity: Whole-Animal Automated Screening for Conserved Fat Regulators. Methods Mol Biol. 2018; 1787: 129–146. https://doi.org/10.1007/978-1-4939-7847-2_10 PMID: 29736715
- Wang MC, O'Rourke EJ, Ruvkun G. Fat metabolism links germline stem cells and longevity in C. elegans. Science. 2008; 322: 957–960. https://doi.org/10.1126/science.1162011 PMID: 18988854
- Long X, Spycher C, Han ZS, Rose AM, Müller F, Avruch J. TOR deficiency in C. elegans causes developmental arrest and intestinal atrophy by inhibition of mRNA translation. Curr Biol. 2002; 12: 1448– 1461. https://doi.org/10.1016/s0960-9822(02)01091-6 PMID: 12225660
- McKay RM, McKay JP, Avery L, Graff JM. C elegans: a model for exploring the genetics of fat storage. Dev Cell. 2003; 4: 131–142. https://doi.org/10.1016/s1534-5807(02)00411-2 PMID: 12530969
- **43.** Sze JY, Victor M, Loer C, Shi Y, Ruvkun G. Food and metabolic signalling defects in a Caenorhabditis elegans serotonin-synthesis mutant. Nature. 2000; 403: 560–564. <u>https://doi.org/10.1038/35000609</u> PMID: 10676966
- 44. Goh GYS, Winter JJ, Bhanshali F, Doering KRS, Lai R, Lee K, et al. NHR-49/HNF4 integrates regulation of fatty acid metabolism with a protective transcriptional response to oxidative stress and fasting. Aging Cell. 2018; 17: e12743. https://doi.org/10.1111/acel.12743 PMID: 29508513
- Van Gilst MR, Hadjivassiliou H, Jolly A, Yamamoto KR. Nuclear hormone receptor NHR-49 controls fat consumption and fatty acid composition in C. elegans. PLoS Biol. 2005; 3: e53. <u>https://doi.org/10. 1371/journal.pbio.0030053</u> PMID: 15719061
- **46.** Lee S-J, Murphy CT, Kenyon C. Glucose shortens the life span of C. elegans by downregulating DAF-16/FOXO activity and aquaporin gene expression. Cell Metab. 2009; 10: 379–391. <u>https://doi.org/10.1016/j.cmet.2009.10.003 PMID: 19883616</u>
- Schlotterer A, Kukudov G, Bozorgmehr F, Hutter H, Du X, Oikonomou D, et al. C. elegans as model for the study of high glucose- mediated life span reduction. Diabetes. 2009; 58: 2450–2456. https:// doi.org/10.2337/db09-0567 PMID: 19675139
- **48.** Lodha D, Rajasekaran S, Jayavelu T, Subramaniam JR. Detrimental effects of fructose on mitochondria in mouse motor neurons and on C. elegans healthspan. Nutr Neurosci. 2020; 1–10. <u>https://doi.org/10.1080/1028415X.2020.1853413</u> PMID: 33258406
- 49. Zheng J, Gao C, Wang M, Tran P, Mai N, Finley JW, et al. Lower Doses of Fructose Extend Lifespan in Caenorhabditis elegans. J Diet Suppl. 2017; 14: 264–277. https://doi.org/10.1080/19390211.2016. 1212959 PMID: 27680107
- Wählby C, Conery AL, Bray M-A, Kamentsky L, Larkins-Ford J, Sokolnicki KL, et al. High- and lowthroughput scoring of fat mass and body fat distribution in C. elegans. Methods. 2014; 68: 492–499. https://doi.org/10.1016/j.ymeth.2014.04.017 PMID: 24784529
- Yengo L, Sidorenko J, Kemper KE, Zheng Z, Wood AR, Weedon MN, et al. Meta-analysis of genomewide association studies for height and body mass index in ~700000 individuals of European ancestry. Hum Mol Genet. 2018; 27: 3641–3649. https://doi.org/10.1093/hmg/ddy271 PMID: 30124842

- Pulit SL, Stoneman C, Morris AP, Wood AR, Glastonbury CA, Tyrrell J, et al. Meta-analysis of genome-wide association studies for body fat distribution in 694 649 individuals of European ancestry. Hum Mol Genet. 2019; 28: 166–174. https://doi.org/10.1093/hmg/ddy327 PMID: 30239722
- Cheng M, Mei B, Zhou Q, Zhang M, Huang H, Han L, et al. Computational analyses of obesity associated loci generated by genome-wide association studies. PLoS One. 2018; 13: e0199987. <u>https://doi.org/10.1371/journal.pone.0199987</u> PMID: 29966015
- Voisin S, Almén MS, Zheleznyakova GY, Lundberg L, Zarei S, Castillo S, et al. Many obesity-associated SNPs strongly associate with DNA methylation changes at proximal promoters and enhancers. Genome Med. 2015; 7: 103. https://doi.org/10.1186/s13073-015-0225-4 PMID: 26449484
- Smemo S, Tena JJ, Kim K-H, Gamazon ER, Sakabe NJ, Gómez-Marín C, et al. Obesity-associated variants within FTO form long-range functional connections with IRX3. Nature. 2014; 507: 371–375. https://doi.org/10.1038/nature13138 PMID: 24646999
- 56. Civelek M, Wu Y, Pan C, Raulerson CK, Ko A, He A, et al. Genetic Regulation of Adipose Gene Expression and Cardio-Metabolic Traits. Am J Hum Genet. 2017; 100: 428–443. https://doi.org/10. 1016/j.ajhg.2017.01.027 PMID: 28257690
- 57. Akiyama M, Okada Y, Kanai M, Takahashi A, Momozawa Y, Ikeda M, et al. Genome-wide association study identifies 112 new loci for body mass index in the Japanese population. Nat Genet. 2017; 49: 1458–1467. https://doi.org/10.1038/ng.3951 PMID: 28892062
- Chu AY, Deng X, Fisher VA, Drong A, Zhang Y, Feitosa MF, et al. Multiethnic genome-wide metaanalysis of ectopic fat depots identifies loci associated with adipocyte development and differentiation. Nat Genet. 2017; 49: 125–130. https://doi.org/10.1038/ng.3738 PMID: 27918534
- 59. Kim W, Underwood RS, Greenwald I, Shaye DD. OrthoList 2: A New Comparative Genomic Analysis of Human and Caenorhabditis elegans Genes. Genetics. 2018; 210: 445–461. <u>https://doi.org/10.1534/ genetics.118.301307 PMID: 30120140</u>
- Kamath RS, Fraser AG, Dong Y, Poulin G, Durbin R, Gotta M, et al. Systematic functional analysis of the Caenorhabditis elegans genome using RNAi. Nature. 2003; 421: 231–237. <u>https://doi.org/10. 1038/nature01278 PMID: 12529635</u>
- Rual J-F, Ceron J, Koreth J, Hao T, Nicot A-S, Hirozane-Kishikawa T, et al. Toward improving Caenorhabditis elegans phenome mapping with an ORFeome-based RNAi library. Genome Res. 2004; 14: 2162–2168. https://doi.org/10.1101/gr.2505604 PMID: 15489339
- Simmer F, Tijsterman M, Parrish S, Koushika SP, Nonet ML, Fire A, et al. Loss of the putative RNAdirected RNA polymerase RRF-3 makes C. elegans hypersensitive to RNAi. Curr Biol. 2002; 12: 1317–1319. https://doi.org/10.1016/s0960-9822(02)01041-2 PMID: 12176360
- Ashrafi K, Chang FY, Watts JL, Fraser AG, Kamath RS, Ahringer J, et al. Genome-wide RNAi analysis of Caenorhabditis elegans fat regulatory genes. Nature. 2003; 421: 268–272. <u>https://doi.org/10.1038/nature01279 PMID</u>: 12529643
- Samuelson AV, Carr CE, Ruvkun G. Gene activities that mediate increased life span of C. elegans insulin-like signaling mutants. Genes Dev. 2007; 21: 2976–2994. <u>https://doi.org/10.1101/gad.1588907</u> PMID: 18006689
- Bachorik JL, Kimble J. Redundant control of the Caenorhabditis elegans sperm/oocyte switch by PUF-8 and FBF-1, two distinct PUF RNA-binding proteins. Proc Natl Acad Sci USA. 2005; 102: 10893– 10897. https://doi.org/10.1073/pnas.0504593102 PMID: 16037210
- Subramaniam K, Seydoux G. Dedifferentiation of primary spermatocytes into germ cell tumors in C. elegans lacking the pumilio-like protein PUF-8. Curr Biol. 2003; 13: 134–139. <u>https://doi.org/10.1016/s0960-9822(03)00005-8 PMID: 12546787</u>
- O'Rourke EJ, Soukas AA, Carr CE, Ruvkun G. C. elegans major fats are stored in vesicles distinct from lysosome-related organelles. Cell Metab. 2009; 10: 430–435. <u>https://doi.org/10.1016/j.cmet.</u> 2009.10.002 PMID: 19883620
- Webster CM, Pino EC, Carr CE, Wu L, Zhou B, Cedillo L, et al. Genome-wide RNAi Screen for Fat Regulatory Genes in C. elegans Identifies a Proteostasis-AMPK Axis Critical for Starvation Survival. Cell Rep. 2017; 20: 627–640. https://doi.org/10.1016/j.celrep.2017.06.068 PMID: 28723566
- Tissenbaum HA, Ruvkun G. An insulin-like signaling pathway affects both longevity and reproduction in Caenorhabditis elegans. Genetics. 1998; 148: 703–717. PMID: <u>9504918</u>
- 70. Fouad AD, Pu SH, Teng S, Mark JR, Fu M, Zhang K, et al. Quantitative Assessment of Fat Levels in Caenorhabditis elegans Using Dark Field Microscopy. G3 (Bethesda). 2017; 7: 1811–1818. <u>https:// doi.org/10.1534/g3.117.040840 PMID: 28404661</u>
- Parks BW, Nam E, Org E, Kostem E, Norheim F, Hui ST, et al. Genetic control of obesity and gut microbiota composition in response to high-fat, high-sucrose diet in mice. Cell Metab. 2013; 17: 141– 152. https://doi.org/10.1016/j.cmet.2012.12.007 PMID: 23312289

- Tappy L, Lê K-A. Metabolic effects of fructose and the worldwide increase in obesity. Physiol Rev. 2010; 90: 23–46. https://doi.org/10.1152/physrev.00019.2009 PMID: 20086073
- Hannou SA, Haslam DE, McKeown NM, Herman MA. Fructose metabolism and metabolic disease. J Clin Invest. 2018; 128: 545–555. https://doi.org/10.1172/JCI96702 PMID: 29388924
- 74. Hanover LM, White JS. Manufacturing, composition, and applications of fructose. Am J Clin Nutr. 1993; 58: 724S–732S. https://doi.org/10.1093/ajcn/58.5.724S PMID: 8213603
- Alcántar-Fernández J, Navarro RE, Salazar-Martínez AM, Pérez-Andrade ME, Miranda-Ríos J. Caenorhabditis elegans respond to high-glucose diets through a network of stress-responsive transcription factors. PLoS One. 2018; 13: e0199888. <u>https://doi.org/10.1371/journal.pone.0199888</u> PMID: 29990370
- 76. Daemen S, van Polanen N, Hesselink MKC. The effect of diet and exercise on lipid droplet dynamics in human muscle tissue. J Exp Biol. 2018; 221. https://doi.org/10.1242/jeb.167015 PMID: 29514886
- 77. Vrablik TL, Petyuk VA, Larson EM, Smith RD, Watts JL. Lipidomic and proteomic analysis of Caenorhabditis elegans lipid droplets and identification of ACS-4 as a lipid droplet-associated protein. Biochim Biophys Acta. 2015; 1851: 1337–1345. https://doi.org/10.1016/j.bbalip.2015.06.004 PMID: 26121959
- 78. Shi X, Li J, Zou X, Greggain J, Rødkær SV, Færgeman NJ, et al. Regulation of lipid droplet size and phospholipid composition by stearoyl-CoA desaturase. J Lipid Res. 2013; 54: 2504–2514. <u>https://doi.org/10.1194/jlr.M039669</u> PMID: 23787165
- 79. Liu Z, Li X, Ge Q, Ding M, Huang X. A lipid droplet-associated GFP reporter-based screen identifies new fat storage regulators in C. elegans. J Genet Genomics. 2014; 41: 305–313. https://doi.org/10. 1016/j.jgg.2014.03.002 PMID: 24894357
- Laranjeiro R, Harinath G, Burke D, Braeckman BP, Driscoll M. Single swim sessions in C. elegans induce key features of mammalian exercise. BMC Biol. 2017; 15: 30. https://doi.org/10.1186/s12915-017-0368-4 PMID: 28395669
- van Hees AMJ, Saris WHM, Hul GB, Schaper NC, Timmerman BE, Lovegrove JA, et al. Effects of dietary fat modification on skeletal muscle fatty acid handling in the metabolic syndrome. Int J Obes. 2010; 34: 859–870. https://doi.org/10.1038/ijo.2010.6 PMID: 20125104
- 82. Britton KA, Fox CS. Ectopic fat depots and cardiovascular disease. Circulation. 2011; 124: e837–41. https://doi.org/10.1161/CIRCULATIONAHA.111.077602 PMID: 22156000
- Yen K, Le TT, Bansal A, Narasimhan SD, Cheng J-X, Tissenbaum HA. A comparative study of fat storage quantitation in nematode Caenorhabditis elegans using label and label-free methods. PLoS One. 2010; 5. https://doi.org/10.1371/journal.pone.0012810 PMID: 20862331
- Soukas AA, Kane EA, Carr CE, Melo JA, Ruvkun G. Rictor/TORC2 regulates fat metabolism, feeding, growth, and life span in Caenorhabditis elegans. Genes Dev. 2009; 23: 496–511. <u>https://doi.org/10. 1101/gad.1775409</u> PMID: 19240135
- Salvestrini V, Sell C, Lorenzini A. Obesity may accelerate the aging process. Front Endocrinol (Lausanne). 2019; 10: 266. https://doi.org/10.3389/fendo.2019.00266 PMID: 31130916
- Ahima RS. Connecting obesity, aging and diabetes. Nat Med. 2009; 15: 996–997. <u>https://doi.org/10.1038/nm0909-996</u> PMID: 19734871
- 87. Tam BT, Morais JA, Santosa S. Obesity and ageing: Two sides of the same coin. Obes Rev. 2020; 21: e12991. https://doi.org/10.1111/obr.12991 PMID: 32020741
- Kimura KD, Tissenbaum HA, Liu Y, Ruvkun G. daf-2, an insulin receptor-like gene that regulates longevity and diapause in Caenorhabditis elegans. Science. 1997; 277: 942–946. https://doi.org/10.1126/ science.277.5328.942 PMID: 9252323
- Halaschek-Wiener J, Khattra JS, McKay S, Pouzyrev A, Stott JM, Yang GS, et al. Analysis of longlived C. elegans daf-2 mutants using serial analysis of gene expression. Genome Res. 2005; 15: 603– 615. https://doi.org/10.1101/gr.3274805 PMID: 15837805
- Hahm J-H, Kim S, DiLoreto R, Shi C, Lee S-JV, Murphy CT, et al. C. elegans maximum velocity correlates with healthspan and is maintained in worms with an insulin receptor mutation. Nat Commun. 2015; 6: 8919. https://doi.org/10.1038/ncomms9919 PMID: 26586186
- Fang Y, McFadden S, Darcy J, Hascup ER, Hascup KN, Bartke A. Lifespan of long-lived growth hormone receptor knockout mice was not normalized by housing at 30°C since weaning. Aging Cell. 2020; 19: e13123. https://doi.org/10.1111/acel.13123 PMID: 32110850
- 92. Fong S, Teo E, Ng LF, Chen C-B, Lakshmanan LN, Tsoi SY, et al. Energy crisis precedes global metabolic failure in a novel Caenorhabditis elegans Alzheimer Disease model. Sci Rep. 2016; 6: 33781. https://doi.org/10.1038/srep33781 PMID: 27653553
- Cherbuin N, Sargent-Cox K, Fraser M, Sachdev P, Anstey KJ. Being overweight is associated with hippocampal atrophy: the PATH Through Life Study. Int J Obes. 2015; 39: 1509–1514. <u>https://doi.org/ 10.1038/ijo.2015.106</u> PMID: 26041696

- 94. Chatterjee S, Peters SAE, Woodward M, Mejia Arango S, Batty GD, Beckett N, et al. Type 2 diabetes as a risk factor for dementia in women compared with men: A pooled analysis of 2.3 million people comprising more than 100,000 cases of dementia. Diabetes Care. 2016; 39: 300–307. https://doi.org/10.2337/dc15-1588 PMID: 26681727
- 95. Yaffe K, Falvey C, Hamilton N, Schwartz AV, Simonsick EM, Satterfield S, et al. Diabetes, glucose control, and 9-year cognitive decline among older adults without dementia. Arch Neurol. 2012; 69: 1170–1175. https://doi.org/10.1001/archneurol.2012.1117 PMID: 22710333
- 96. Teo E, Ravi S, Barardo D, Kim H-S, Fong S, Cazenave-Gassiot A, et al. Metabolic stress is a primary pathogenic event in transgenic Caenorhabditis elegans expressing pan-neuronal human amyloid beta. Elife. 2019; 8. https://doi.org/10.7554/eLife.50069 PMID: 31610847
- Mitchell P, Mould R, Dillon J, Glautier S, Andrianakis I, James C, et al. A differential role for neuropeptides in acute and chronic adaptive responses to alcohol: behavioural and genetic analysis in Caenorhabditis elegans. PLoS One. 2010; 5: e10422. https://doi.org/10.1371/journal.pone.0010422 PMID: 20454655
- 98. Gallagher MD, Chen-Plotkin AS. The Post-GWAS Era: From Association to Function. Am J Hum Genet. 2018; 102: 717–730. https://doi.org/10.1016/j.ajhg.2018.04.002 PMID: 29727686
- Srinivasan S. Regulation of body fat in Caenorhabditis elegans. Annu Rev Physiol. 2015; 77: 161– 178. https://doi.org/10.1146/annurev-physiol-021014-071704 PMID: 25340962
- 100. Srinivasan S, Sadegh L, Elle IC, Christensen AGL, Faergeman NJ, Ashrafi K. Serotonin regulates C. elegans fat and feeding through independent molecular mechanisms. Cell Metab. 2008; 7: 533–544. https://doi.org/10.1016/j.cmet.2008.04.012 PMID: 18522834
- Kurzchalia TV, Ward S. Why do worms need cholesterol? Nat Cell Biol. 2003; 5: 684–688. https://doi. org/10.1038/ncb0803-684 PMID: 12894170
- 102. Uno M, Nishida E. Lifespan-regulating genes in C. elegans. npj Aging Mech Dis. 2016; 2: 16010. https://doi.org/10.1038/npjamd.2016.10 PMID: 28721266
- 103. Kenyon C. The first long-lived mutants: discovery of the insulin/IGF-1 pathway for ageing. Philos Trans R Soc Lond B, Biol Sci. 2011; 366: 9–16. https://doi.org/10.1098/rstb.2010.0276 PMID: 21115525
- 104. Brina D, Miluzio A, Ricciardi S, Clarke K, Davidsen PK, Viero G, et al. eIF6 coordinates insulin sensitivity and lipid metabolism by coupling translation to transcription. Nat Commun. 2015; 6: 8261. https:// doi.org/10.1038/ncomms9261 PMID: 26383020
- 105. Sun X, Fu X, Li J, Xing C, Martin DW, Zhang HH, et al. Heterozygous deletion of Atbf1 by the Cre-loxP system in mice causes preweaning mortality. Genesis. 2012; 50: 819–827. <u>https://doi.org/10.1002/dvg.22041 PMID: 22644989</u>
- 106. Lin H-P, Cheng Z-L, He R-Y, Song L, Tian M-X, Zhou L-S, et al. Destabilization of fatty acid synthase by acetylation inhibits de novo lipogenesis and tumor cell growth. Cancer Res. 2016; 76: 6924–6936. https://doi.org/10.1158/0008-5472.CAN-16-1597 PMID: 27758890
- 107. Oh K-J, Park J, Kim SS, Oh H, Choi CS, Koo S-H. TCF7L2 modulates glucose homeostasis by regulating CREB- and FoxO1-dependent transcriptional pathway in the liver. PLoS Genet. 2012; 8: e1002986. https://doi.org/10.1371/journal.pgen.1002986 PMID: 23028378
- 108. Ip W, Shao W, Song Z, Chen Z, Wheeler MB, Jin T. Liver-specific expression of dominant-negative transcription factor 7-like 2 causes progressive impairment in glucose homeostasis. Diabetes. 2015; 64: 1923–1932. https://doi.org/10.2337/db14-1329 PMID: 25576056
- 109. Abranches MV, Oliveira FCE de, Conceição LL da, Peluzio M do CG. Obesity and diabetes: the link between adipose tissue dysfunction and glucose homeostasis. Nutr Res Rev. 2015; 28: 121–132. https://doi.org/10.1017/S0954422415000098 PMID: 26650242
- 110. Brooks KK, Liang B, Watts JL. The influence of bacterial diet on fat storage in C. elegans. PLoS One. 2009; 4: e7545. https://doi.org/10.1371/journal.pone.0007545 PMID: 19844570
- 111. Stuhr NL, Curran SP. Bacterial diets differentially alter lifespan and healthspan trajectories in C. elegans. Commun Biol. 2020; 3: 653. https://doi.org/10.1038/s42003-020-01379-1 PMID: 33159120
- 112. Neve IAA, Sowa JN, Lin C-CJ, Sivaramakrishnan P, Herman C, Ye Y, et al. Escherichia coli Metabolite Profiling Leads to the Development of an RNA Interference Strain for Caenorhabditis elegans. G3 (Bethesda). 2020; 10: 189–198. https://doi.org/10.1534/g3.119.400741 PMID: 31712257
- 113. Han B, Sivaramakrishnan P, Lin C-CJ, Neve IAA, He J, Tay LWR, et al. Microbial genetic composition tunes host longevity. Cell. 2017; 169: 1249–1262.e13. <u>https://doi.org/10.1016/j.cell.2017.05.036</u> PMID: 28622510
- 114. Johanns M, Kviklyte S, Chuang S-J, Corbeels K, Jacobs R, Herinckx G, et al. Genetic deletion of soluble 5'-nucleotidase II reduces body weight gain and insulin resistance induced by a high-fat diet. Mol Genet Metab. 2019; 126: 377–387. https://doi.org/10.1016/j.ymgme.2019.01.017 PMID: 30803894

- 115. Mannan AU, Roussa E, Kraus C, Rickmann M, Maenner J, Nayernia K, et al. Mutation in the gene encoding lysosomal acid phosphatase (Acp2) causes cerebellum and skin malformation in mouse. Neurogenetics. 2004; 5: 229–238. https://doi.org/10.1007/s10048-004-0197-9 PMID: 15503243
- 116. Branen JK, Shintani DK, Engeseth NJ. Expression of antisense acyl carrier protein-4 reduces lipid content in Arabidopsis leaf tissue. Plant Physiol. 2003; 132: 748–756. <u>https://doi.org/10.1104/pp.102.018622</u> PMID: 12805604
- 117. Jura M, Kozak LP. Obesity and related consequences to ageing. Age (Omaha). 2016; 38: 23. https:// doi.org/10.1007/s11357-016-9884-3 PMID: 26846415
- Klarin D, Damrauer SM, Cho K, Sun YV, Teslovich TM, Honerlaw J, et al. Genetics of blood lipids among ~300,000 multi-ethnic participants of the Million Veteran Program. Nat Genet. 2018; 50: 1514– 1523. https://doi.org/10.1038/s41588-018-0222-9 PMID: 30275531
- 119. Brewer HB. High-density lipoproteins: a new potential therapeutic target for the prevention of cardio-vascular disease. Arterioscler Thromb Vasc Biol. 2004; 24: 387–391. <u>https://doi.org/10.1161/01.ATV.0000121505.88326.d2</u> PMID: 15003970
- 120. Lima D, Machado A, Reis-Henriques MA, Rocha E, Santos MM, Castro LFC. Cloning and expression analysis of the 17β hydroxysteroid dehydrogenase type 12 (HSD17B12) in the neogastropod Nucella lapillus. J Steroid Biochem Mol Biol. 2013; 134: 8–14. <u>https://doi.org/10.1016/j.jsbmb.2012.10.005</u> PMID: 23069646
- 121. Rantakari P, Lagerbohm H, Kaimainen M, Suomela J-P, Strauss L, Sainio K, et al. Hydroxysteroid (17 {beta}) dehydrogenase 12 is essential for mouse organogenesis and embryonic survival. Endocrinology. 2010; 151: 1893–1901. https://doi.org/10.1210/en.2009-0929 PMID: 20130115
- 122. Stiernagle T. Maintenance of C. elegans. WormBook. 2006; 1–11. https://doi.org/10.1895/wormbook. 1.101.1 PMID: 18050451
- 123. Wählby C, Kamentsky L, Liu ZH, Riklin-Raviv T, Conery AL, O'Rourke EJ, et al. An image analysis toolbox for high-throughput C. elegans assays. Nat Methods. 2012; 9: 714–716. <u>https://doi.org/10.1038/nmeth.1984 PMID: 22522656</u>
- 124. Nussbaum-Krammer CI, Neto MF, Brielmann RM, Pedersen JS, Morimoto RI. Investigating the spreading and toxicity of prion-like proteins using the metazoan model organism C. elegans. J Vis Exp. 2015; 52321. https://doi.org/10.3791/52321 PMID: 25591151
- 125. Podbilewicz B, Gruenbaum Y. Live Imaging of Caenorhabditis elegans: preparation of samples. CSH Protoc. 2006; 2006. https://doi.org/10.1101/pdb.prot4601 PMID: 22485988